www.cambridge.org/pab

Invited Article

Cite this article: Blois, J. L., A. M. Bellvé, M. A. Jarzyna, E. E. Saupe, and V. J. P. Syverson (2025). Paleobiogeographic insights gained from ecological niche models: progress and continued challenges. *Paleobiology*, 1–21 https://doi.org/10.1017/pab.2024.16

Received: 04 December 2023 Revised: 15 April 2024 Accepted: 26 April 2024

Handling Editor: Mark Patzkowsky

Corresponding author: Jessica L. Blois; Email: jblois@ucmerced.edu

© The Author(s), 2025. Published by Cambridge University Press on behalf of Paleontological Society. This is an Open Access article, distributed under the terms of the Creative Commons Attribution licence (http:// creativecommons.org/licenses/by/4.0), which permits unrestricted re-use, distribution and reproduction, provided the original article is properly cited.



Paleobiogeographic insights gained from ecological niche models: progress and continued challenges

Jessica L. Blois¹, André M. Bellvé², Marta A. Jarzyna³, Erin E. Saupe⁴, and Val J. P. Syverson⁵

¹Sierra Nevada Research Institute and Department of Life and Environmental Sciences, University of California–Merced, Merced, California, U.S.A.

²Department of Evolution, Ecology and Organismal Biology, The Ohio State University, Columbus, Ohio, U.S.A.
³Department of Evolution, Ecology and Organismal Biology and Translational Data Analytics Institute, The Ohio State University, Columbus, Ohio, U.S.A.

⁴Department of Earth Sciences, University of Oxford, Oxford, U.K.

⁵Department of Life and Environmental Sciences, University of California–Merced, Merced, California, U.S.A.

Abstract

The spatial distribution of individuals within ecological assemblages and their associated traits and behaviors are key determinants of ecosystem structure and function. Consequently, determining the spatial distribution of species, and how distributions influence patterns of species richness across ecosystems today and in the past, helps us understand what factors act as fundamental controls on biodiversity. Here, we explore how ecological niche modeling has contributed to understanding the spatiotemporal distribution of past biodiversity and past ecological and evolutionary processes. We first perform a semiquantitative literature review to capture studies that applied ecological niche models (ENMs) to the past, identifying 668 studies. We coded each study according to focal taxonomic group, whether and how the study used fossil evidence, whether it relied on evidence or methods in addition to ENMs, spatial scale of the study, and temporal intervals included in the ENMs. We used trends in publication patterns across categories to anchor discussion of recent technical advances in niche modeling, focusing on paleobiogeographic ENM applications. We then explored contributions of ENMs to paleobiogeography, with a particular focus on examining patterns and associated drivers of range dynamics; phylogeography and within-lineage dynamics; macroevolutionary patterns and processes, including niche change, speciation, and extinction; drivers of community assembly; and conservation paleobiogeography. Overall, ENMs are powerful tools for elucidating paleobiogeographic patterns. ENMs are most commonly used to understand Quaternary dynamics, but an increasing number of studies use ENMs to gain important insight into both ecological and evolutionary processes in pre-Quaternary times. Deeper integration with traits and phylogenies may further extend those insights.

Non-technical Summary

The spatial distribution of species across the landscape and their associated traits and behaviors play a pivotal role in determining ecosystem structure and function and contribute to our understanding of the processes that shape biodiversity. Ecological niche models (ENMs) are tools that can be used to estimate the ecological niche of a species based on its known occurrences. In this review, we explore the ways that ENMs have been used to study the evolution and ecology of past biodiversity. While ENMs are commonly used to understand the dynamics of species and assemblages during more recent periods of Earth history (i.e., the last several million years), an increasing number of studies have extended ENMs deeper into the geologic past. Overall, ENMs are powerful tools for illuminating paleobiogeographic patterns; further integration of ENMs with traits, phylogenies, and other methods may extend insights.

Introduction

The spatial distribution of individuals within ecological assemblages and their associated traits and behaviors are key determinants of the structure and function of ecosystems and, ultimately, the services they provide, which support life on Earth (e.g., Cardinale et al. 2012; Tilman et al. 2014; van der Plas 2019). Thus, biodiversity is, and has always been, a key regulator of planetary homeostasis (Mace et al. 2014; Steffen et al. 2018; Talukder et al. 2022). Understanding where and why different species exist—now, in the past, and in the future—reveals how that regulation operates and how humans have altered it (Lyons et al. 2016; Barnosky et al. 2017). Determining



the spatial distribution of different species, and how distributions influence the spatial patterns of species richness across different ecosystems today and in the past, helps us understand what factors act as fundamental controls on biodiversity.

Where species occur reflects aspects of their environment (e.g., Grinnell 1917; Hutchinson 1957; Colwell and Rangel 2009; Soberón and Nakamura 2009). This correspondence between species and environment has been used to understand the factors supporting species' persistence (e.g., Holt 2009; Scheele et al. 2017), to examine the interactions between species (e.g., Wiens 2011; Blois et al. 2013), and to infer past climates and other environmental conditions (Fagoaga et al. 2019; Chevalier et al. 2020; Wei et al. 2020). Today, this relationship is increasingly being used to predict potential future ranges of species, given global anthropogenic climate change (Dietl and Flessa 2011; Kuemmerle et al. 2012; Lima-Ribeiro et al. 2017; Ivory et al. 2019).

Since the inception of Paleobiology 50 years ago, and particularly over the last few decades, scientists have used ecological niche models (ENMs) to define species-environment relationships. We expand on definitions of the niche, our use of the term "ENMs", and related terms in Box 1. In short, ENMs are statistical models in which species' occurrences are related to different aspects of the climate or resources (collectively, "the environment") and are used to infer species' niches (Guisan and Thuiller 2005; Elith and Leathwick 2009; Holt 2009; Fig. 1). Species' occurrences are sourced from field surveys, primary literature, museum records, or often through databases that aggregate these occurrences, such as the Global Biodiversity Information Facility (GBIF; https://www.gbif.org), the Neotoma Paleoecology Database (https://www.neotomadb.org; Williams et al. 2018), and the Paleobiology Database (PBDB; https://paleobiodb.org). This correlative approach does not guarantee that all factors supporting a species' persistence (i.e., its fundamental niche) will be captured.

However, by statistically characterizing the relationships between species' occurrences and the environment, it is possible to identify a subset of environmental factors that are strongly associated with species' presence (Fig. 1A) and thus approximate the species' niche (i.e., its "realized" niche; Fig. 1B). This statistical relationship allows us to investigate why species are found in some places or environments and not others (Fig. 1C), providing better insights into the fundamental controls on species' distributions and enabling forecasting of future distributions.

A variety of past reviews have illustrated that inclusion of data collected across different times and environments, particularly from past environments without any contemporary analogue (i.e., paleoecological niche models [paleoENMs]; Box 1), allows for more complete characterization of the niche and thus better understanding of paleobiogeography (Nógues-Bravo 2009; Maguire et al. 2015; Myers et al. 2015; Lima-Ribeiro et al. 2017). PaleoENMs are not a panacea and indeed are subject to a variety of challenges akin to those encountered in ecological niche modeling more broadly (e.g., Guisan and Thuiller 2005; Elith and Leathwick 2009; Saupe et al. 2012). For example, limited fossil occurrences may exacerbate the issue of low sample size in paleoENMs, while taphonomic biases may make it difficult to interpret data on species' absences. Furthermore, paleoecological niche modeling is typically not possible for taxa that do not readily fossilize. In addition, environmental layers are typically lower resolution and not as easily obtainable for time periods of the past as they are for the present day (Nógues-Bravo 2009; Svenning et al. 2011; Varela et al. 2011; Maguire et al. 2015; Myers et al. 2015), which has prompted the development of alternative methods based on sedimentary and stratigraphic characteristics (Stigall 2023; Holland et al. 2024). Despite these challenges, niche modeling is an extremely useful tool for paleobiogeography, and better integration of fossil

Box 1. Definitions of terms

The concept of a niche (sensu Chase and Leibold 2003) encompasses both the set of resources that support population stability and/or growth (the "Grinnellian" niche; Soberón 2007) as well as the per capita impacts that a species has on those resources (the "Eltonian" niche; Soberón 2007). Although acknowledging both, our primary focus in this paper is on the Grinnellian niche, which we will refer to simply as "the niche" henceforth.

Understanding species' niches has been a pursuit spanning centuries. The definition of the Grinnellian niche was formalized by Hutchinson (1957) as an *n*dimensional hypervolume, typically characterized by multiple environmental factors that support species' persistence. The *n*-dimensional hypervolume is now commonly used to examine spatiotemporal patterns in species' niches and distributions, although defining species' requirements and tolerances is simple in concept but often difficult in practice. Today, ecological niche models (ENMs) play a crucial role in estimating species' niches by defining species-environment relationships. ENMs are statistical models, wherein current species' occurrences (e.g., modern-day presence, presence/absence, abundance) are related to diverse environmental factors, such as contemporary climate, resources, and other relevant variables, collectively termed "the environment" (Fig. 1A). ENMs come in a variety of forms, encompassing both parametric and nonparametric approaches.

Under a parametric framework, ENMs commonly employ regression-type analysis. Here, species occurrences serve as the dependent variable, while environmental factors such as climate and vegetation act as independent variables within the model equation. The resultant statistical relationships, depicted as response curves, delineate the species' niche within a multidimensional environmental space defined by environmental variables included in the ENM (Fig. 1B). These curves convey the probability of events, such as a species' occurrence, at distinct points within the environmental space. Within a nonparametric framework, methods like kernel density estimation (KDE) can be utilized to build ENMs. KDE generates a continuous probability density function (PDF) based on a finite set of observations (i.e., species' occurrences) along one or more independent environmental axes. Integrating this PDF over specified intervals yields the probability of events occurring within those environmental ranges. Both parametric and nonparametric methods can be employed across one or more environmental dimensions.

While ENMs are calibrated on present-day data, we define paleoecological niche models (paleoENMs) as ENMs trained on paleo-occurrences and corresponding paleoenvironmental layers. In essence, paleoENMs leverage the fossil record for species' occurrence data and utilize past environmental data that align spatially and temporally with the fossil record to obtain past species—environment relationships. Note that, while the paleoenvironmental data incorporated into paleoENMs can derive from various sources (see "Environmental Reconstruction" section), we do not differentiate between these data types but rather consider them collectively as paleoenvironmental information. In both ENMs and paleoENMs, model validation typically involves evaluating the model's performance by comparing its predictions against an independent dataset.

Species–environment relationships estimated with ENMs or paleoENMs can then be used to project environmental suitability or species probability of occurrence within a multidimensional environmental space or across geographic space (Fig. 1C). The term "projection" encompasses forecasts, hindcasts, and projection to contemporaneous times. Forecasts are projections of niche models to time periods subsequent to the period in which the ENM or paleoENM was developed, while hindcasts are projections to time periods preceding the interval for which the ENM or paleoENM was developed. We note that if projections are made in geographic space, then the ENM could become a species distribution model (SDM). While the terms SDM and ENM are often used interchangeably in the literature, we consider SDMs to be a subset of ENMs, with the explicit intent of modeling geographic distributions, which often requires additional information on a species' dispersal potential and biotic factors, while defining ecological niches via ENMs does not require these as inputs.



Figure 1. Ecological (ENMs) and paleoecological (paleoENMs) niche models integrate (A) species occurrence data with environmental layers to obtain (B) characterizations of species niches within an *n*-dimensional environmental space across time. Those niches are then (C) projected either contemporaneously or through hindcast (before the time interval for which ENM/paleoENM was developed) and forecast (subsequent to the time interval for which ENM/paleoENM was developed) projections to assess habitat suitability either in the original niche space or in geographic space. For a more accurate representation of species' fundamental niches, (D) aggregating occurrences across multiple time periods generates pooled niches that can be used for projections into distinct time intervals.

data into ENMs has the potential to provide a deeper understanding of species' niches, species distributions, and past biogeographic patterns.

Here, we explore how ecological niche modeling has been used to understand the spatiotemporal distribution of biodiversity in the past. The conceptual framework underlying ENMs and paleoENMs



Figure 2. Stacked bar chart of the number (#) of publications that applied ecological niche models to past time intervals, ordered by publication year and categorized and colored by the authors' use of fossils. To facilitate comparison of change in the proportion of publications that did not rely on fossil data in any way, the values listed above the bars show the rounded percentage of publications in the "No Fossils" category for each year.

has been examined previously (Guisan and Thuiller 2005; Soberón and Nakamura 2009; Myers et al. 2015), and a series of excellent reviews has highlighted the strengths and challenges of modeling past species' distributions and niches (Nógues-Bravo 2009; Svenning et al. 2011; Varela et al. 2011; McGuire and Davis 2014; Maguire et al. 2015; Myers et al. 2015; Moreno-Amat et al. 2017) and the paucity of studies incorporating fossil data into ENMs (Nógues-Bravo 2009; Svenning et al. 2011; Varela et al. 2011). We first assess the state of paleo-niche modeling through a semiquantitative literature review (SQLR), asking whether progress has been made in the past 10+ years in addressing previously outlined challenges of paleoENMs. We next discuss advances in technical development of paleoENMs, and then move toward a detailed overview of how paleoENMs have advanced our understanding of ecological and evolutionary patterns and processes. Finally, we explore emerging frontiers in niche-modeling approaches to paleobiogeography.

Semiquantitative Literature Review

To anchor our review, we conducted a search on 15 September 2023 for peer-reviewed articles, written in English, that applied ENMs to past time intervals using both the Scopus and Web of Science databases with near-identical search conditions (see Supplementary Appendix 1 for full search terms). Our search and screening followed the PRISMA protocol for scoping reviews (Tricco et al. 2018). Article metadata were downloaded from each database (Scopus n = 16,155, Web of Science n = 15,600), and the two datasets were merged and duplicates removed (n = 22,656). We screened article titles and abstracts to determine if they (1) projected an ENM to a point in time before 1800 C.E. and/or (2) included fossil

occurrences in their ENM. We identified 668 studies that met our criteria and randomly assigned these to the five authors to gather data on the ENM approaches therein. Data extracted from each article included taxonomic information (taxonomic description and resolution, and the number of taxonomic units analyzed), time periods for which data were modeled and projected, use of the fossil record for either model calibration or validation, additional data (molecular, isotopic, morphological, etc.) used, and the geographic extent of the analysis. All data manipulation and analyses were performed in R (v. 4.3.0; R Core Team 2014) using an RStudio interface (v. 2023.06.1 Build 524 "Mountain Hydrangea"; RStudio Team 2020). Data manipulations were carried out with dplyr (v. 1.1.2; Wickham et al. 2023a), tidyr (v. 1.3.0; Wickham et al. 2023b), and stringr (v. 1.5.0; Wickham 2023). Title and abstract screening was done through revtools (v. 0.4.1; Westgate 2019).

Fossil evidence can be used in one of two ways in nichemodeling approaches: as occurrences during model training to infer the ecological niche and (potentially) geographic distributions (defined here as paleoENMs; Box 1) and/or to validate hindcast or forecast projections (Box 1, Fig. 1). We found that while the number of studies that apply ENMs and paleoENMs to understand past niches and distributions has increased and niche modeling is now a common tool in many different facets of paleobiogeography, ENMs that rely on fossil evidence in some way are still proportionately rare and appear to have reached a "steady state" (Fig. 2). Slightly more studies employing ENMs for paleobiogeography focus on animals than on plants, and a higher proportion of animal-based ENMs or paleoENMs incorporate fossil evidence into the niche models versus hindcasting ENMs to the past from contemporary occurrences (Fig. 3A). Most ENMs developed for paleobiogeographic



Figure 3. Stacked bar chart of the number (#) of publications that applied ecological niche models to past time intervals, ordered by publication year and showing (A) the number of publications focused on the Animalia and Plantae kingdoms, and (B) the number of publications that combined ecological niche models (ENMs) with other lines of evidence or where ENMs were the sole approach used in the study.

studies are combined with some other approach, but studies that use just ENMs or paleoENMs are more likely to incorporate fossil evidence into their methods (Fig. 3B). The vast majority of paleobiogeographic studies that use niche modeling are focused on the regional scale and concerned with the Quaternary (Figs. 4, 5), although there has been a steady trend toward using niche-modeling approaches in paleobiogeography at older times (Fig. 5B). We examine these patterns in more depth in the subsequent sections.

Advances in PaleoENM Methods

Several key reviews have highlighted the potential of paleoENMs in paleobiogeography (Dietl and Flessa 2011; Svenning et al. 2011; Varela et al. 2011; Franklin et al. 2015; Maguire et al. 2015; Myers et al. 2015; Moreno-Amat et al. 2017), but these reviews also illustrated some key concerns of ENMs and paleoENMs that may limit their utility. Here, we examine several key areas related to the technical development of paleoENMs that have seen major advances in recent years and that are particularly relevant to paleobiogeography:

taphonomic biases, small sample sizes and taxonomy, environmental reconstructions, and model transferability.

Taphonomic Bias

Fossils can only tell us about the environments in which they can be preserved, and thus do not generally provide us with true absence information (Franklin et al. 2015; Moreno-Amat et al. 2017). Fossils are also subject to many of the same sampling biases that afflict contemporary occurrence data, such as uneven search effort through space and time (Inman et al. 2018, 2021) or environmental influences on detection (Baker et al. 2022). The cumulative effect of these biases, if not accounted for, can result in misleading or incomplete estimations of species' niches. Accuracy can be improved by accounting for the probability of fossil discovery and preservation. For example, Block et al. (2016) demonstrated increased predictive accuracy of ENMs for late Quaternary megafauna by incorporating sampling bias covariates, which they obtained by modeling the occurrence of all late Quaternary



Figure 4. Stacked bar chart of the number (#) of publications that applied ecological niche models (ENMs) to past time intervals, ordered by publication year and categorized and colored by their use of fossils, showing (A) the number of publications focused on different geographic scales; (B), the number of publications focused on different geologic era; (C), the number of publications focused on different geologic periods (excluding the Quaternary); and (D), the number of publications focused on different time intervals within the Quaternary: Holocene, 11.7–0 ka; LGM – deglacial, 22–11.7 ka; LIG – LGM, 140 – 22ka; early Pleistocene, 2.6 Ma–140 ka. LGM: last glacial maximum; LIG: last interglacial.

megafaunal fossils as functions of environmental features known to be related to preservation or discovery. An analogous method used in contemporary ENM approaches is weighting pseudo-absence point selection by sampling effort (e.g., distance to road), a practice sometimes used for records drawn from databases such as GBIF (Phillips et al. 2009; Inman et al. 2021). When pseudo-absences are selected at random, there is a higher chance that environments where a species is actually present, but not observed, are erroneously coded as absent. By weighting pseudo-absence selection to emphasize areas that are more likely to have been sampled, we are more likely to capture true absences. Inman et al. (2018) extended this approach to paleoENMs by creating three separate statistical models that accounted for the availability, preservation, and discovery of fossil data, the product of which was then used to weight pseudo-absence selection. Weighting pseudo-absence point selection for paleoENMs was subsequently used by Lentini et al. (2018) and Jarvie et al. (2021) to model the prehuman distribution of kākāpō (Strigops habroptilus) and tuatara (Sphenodon punctatus), respectively. In some cases, absence data-or more accurate inference of pseudo-absences-may also be inferred through use of taphonomic control taxa (Bottjer and Jablonski 1988; Jablonski

et al. 1997; Behrensmeyer et al. 2000). In this case, the lack of fossil occurrences is assumed to record "true" absence when taxa that have similar ecological and depositional characteristics and biases have been found in the same assemblage. For example, Veloz et al. (2012) treated the lack of fossil pollen observations as indicative of true absence when developing their ENMs, because the pollen taxa in their study were "readily identifiable by palynologists" (p. 1700) and experienced similar depositional biases. Overall, paleoENM research would benefit from standardized methods for taphonomic and discovery bias layer creation and a more explicitly quantified model of preservation.

Sample Size and Taxonomy

Preservation processes might also lead to other less tractable challenges. For instance, paleoENMs are often hindered by small sample sizes and unresolved taxonomies, both of which may vary among taxa, time, or space. As fossilization favors particular ecological niches and the recent past, sample sizes may be unevenly distributed across environments and time, distorting our perception of a species' niche. In addition, identifying taxa from fossils is



Publication Year

Figure 5. Trends across publication year in the times to which ENMs or paleoENMs are projected, focused on the subset of studies that project to (A) the last 160 kyr and (B) older periods in Earth history (0.5–100 Ma). Note that due to sparse data, we do not show the subset of studies that project to times older than 100 Ma. Studies that used fossils for either model development or model validation are shown on the left, and studies that did not rely on fossil evidence are shown on the right. Each color plus symbol combination shows the type of model (Fig. 1). Three time periods with widely available environmental layers are indicated on the right. For example, studies that incorporate fossil evidence can be either hindcast, forecast, or projected to contemporaneous times, whereas all studies that did not use fossils are necessarily hindcast to the past from the present. MH, mid-Holocene; LGM, last glacial maximum; LIG, last interglacial.

often not possible at the species level, especially for smaller-bodied species, species known only from limited fossil fragments, or the pollen of many plant taxa, further diminishing sample sizes. In fact,

the inability to identify some plant species from pollen led Moreno-Amat et al. (2017) to suggest that paleoENMs should be limited to monospecific genera or to localities where taxa can be resolved to the species level; for example, plant macrofossil data that verify species' presences suggested by genus-level fossil pollen taxa, or study areas without overlap in ambiguous pollen taxa. Challenges in species identification in palynology may be a key driver in differential use of fossils in ENMs between the Animalia and Plantae kingdoms (Fig. 3A). For example, relatively few studies incorporate plant fossil data (vs. animal fossils) into ENMs that are hindcast to, for example, the mid-Holocene 6 ka or last glacial maximum (LGM) ~21 ka, despite the availability of late Quaternary occurrences for both fossil pollen and mammals during this time (Williams et al. 2018), perhaps because fossil pollen data are frequently identified to the genus or higher level (Ritchie 1995; Moreno-Amat et al. 2017).

One possible solution to these issues may be to aggregate occurrences of ecologically similar taxa. Merging of records, particularly of closely related species, has been done in numerous studies (e.g., Waterson et al. 2016; Brooke et al. 2022), perhaps most notably for paleoENMs of biomes (e.g., Roberts and Hamann 2012; Tarkhnishvili et al. 2012; Werneck et al. 2012; Hope et al. 2013; Arruda et al. 2018); this approach is also common in other studies using paleontological data (Valentine 1969; Hadly et al. 2009; Patzkowsky 2017). Additionally, higher taxa are often used as a proxy for species due to taxonomic uncertainties and resolution in studies involving the fossil record. Examining niches at higher taxonomic levels may be relevant due to niche conservatism, wherein closely related species share similar ecological characteristics (e.g., Wiens et al. 2010). While niche differences may exist among species within a higher taxon, these differences may not be distinguishable at the spatial or temporal resolution of the ENM or relevant to the kind of niche being defined (e.g., Jackson et al. 2009). In these cases, additional fossil occurrence data from ecologically similar species may supplement small sample sizes, although they should be considered with care (Hendricks et al. 2014). In support of merging occurrences across taxa, Qiao et al. (2017) found predictive performance of ENMs based on virtual species improved when ecologically similar species were combined, irrespective of evaluation metric or algorithm used. Although defining ecological similarity may be difficult, established methods for evaluating niche overlap (e.g., Broennimann et al. 2012) could be leveraged. Merging taxa for use in ENMs does mean that resulting ENMs should be interpreted with caution (e.g., see Hendricks et al. 2014) and only at the taxonomic resolution used and for the kind of niche being modeled, with explicit caveats included by the authors.

Environmental Reconstruction

ENMs are reliant on two sets of input data: taxon occurrences and inferences of the environment associated with the occurrences (Fig. 1). For ENMs projected to the past or developed using paleodata, environmental layers typically are drawn from several different sources, including (1) Earth system models that provide estimates of climate during different snapshots of time in the past; (2) environmental reconstructions or proxies drawn from associated sedimentary, stratigraphic, or geochemical archives; and (3) environmental proxies derived from ecometric associations (see Varela et al. [2010], Myers et al. [2015], and Lawing [2021] for discussion of commonly used and emerging paleoenvironmental layers). To some extent, reliance on different types of environmental information is related to timescale; deeper-time ENMs (i.e., those that model pre-Quaternary taxa) typically rely on environmental reconstructions from sedimentary proxies, while more recent ENMs (for contemporary or Quaternary taxa) typically rely on paleoclimate models (Myers et al. 2015). Regardless of environmental

reconstruction, numerous issues can influence the fit or interpretation of niche models, including the extent of time averaging, potential mismatches in the spatial and temporal resolution between occurrences and environmental reconstructions, the extent of uncertainty due to extrapolation among few data points, and limited availability of modeled or reconstructed environmental variables (see Myers et al. [2015] for a fuller discussion). Although these issues can affect ENMs developed for any time period, including contemporary ENMs, they are typically more challenging or limiting for paleoENMs.

The availability of rich paleoenvironmental information has increased greatly in recent years. Development of new isotopic proxies and efforts to compile databases of sedimentological and geochemical data have expanded the information extracted from sedimentary archives (e.g., Janus [Mithal and Becker 2006]; StabisoDB [stabisodb.org], Macrostrat [Peters et al. 2018]). Additionally, Earth system model output is available for an expanding number of time intervals, which provides another source of environmental reconstruction for deep-time ENMs. For example, the Paleoclimate Model Intercomparison Project (PMIP) has iteratively expanded the temporal scope of deeper-time paleoclimate models, which are then adapted for use in niche modeling. PMIP1 modeled the mid-Holocene and LGM as core modeling targets, but the last interglacial, Pliocene, and older times (DEEP, targeting times in the Eocene and Miocene) were not added as "core" intercomparison modeling projects until PMIP3 or PMIP4 (though paleoclimates of some of those times were modeled by smaller groups at earlier stages) (Braconnot et al. 2021). Paleoclimate layers are now available for each stage in the Cretaceous through the Eocene (Farnsworth et al. 2019), for each 10 kyr interval since the Pliocene (Lima-Ribeiro et al. 2015; Brown et al. 2018; Gamisch 2019), and for a variety of intervals in the late Quaternary (e.g., Fordham et al. 2017; Karger et al. 2023). The broader availability of paleoclimate models can be seen in our SQLR (Fig. 5): ENMs built for or projected to the last interglacial period ca. 125 ka first appeared in 2010 and became more common (deeper shades of purple) in the mid-2010s, with more time intervals within the late Quaternary now available for niche modeling (vertical strips of symbols). Similarly, a very clear expansion of ENMs projected to pre-Quaternary times is apparent throughout the 2010s (Fig. 5).

The environment can also be inferred using stratigraphic paleobiology (Holland et al. 2024) to "reconstruct" the primary axes of environmental variation relevant to taxa. In this approach, the primary axes of variation among the taxa present within the entire fossil assemblage are deduced using ordination methods such as detrended correspondence analysis or nonmetric multidimensional scaling, then integrated with a detailed stratigraphic model of the study region to infer the relevant environmental parameters structuring the assemblage. Although the approach does not produce ENMs as we define them here, niche parameters for individual taxa can then be inferred from the resulting ecospace and used to analyze niche change or stability (see "Niche Change" section).

Model Transferability

The SQLR results illustrate that, in many studies, ENMs and paleoENMs were developed, at least in part, to project species' niches to other time periods (Fig. 5). In the majority of papers we reviewed, projections were hindcast from models developed without relying on fossil data, and the hindcasts were not validated using fossil data (Fig. 5). However, ENMs are largely beholden to the occurrences used to train the models, and realized niches based on

contemporary data tell only part of the story. When trained solely on contemporary occurrences, ENMs tend to underestimate the breadth of a species' Grinnellian niche, and therefore the extent of suitable conditions for a species (Hortal et al. 2008). In addition, and perhaps crucially, the set of environmental factors and their correlations with one another across the landscape is different today than in even the recent past (Jackson and Overpeck 2000). Together, these factors mean that, although models may accurately predict the contemporaneous niche of a species, they may have limited temporal transferability; that is, they may not accurately predict past and future niches or distributions (Svenning et al. 2011; Varela et al. 2011; Davis et al. 2014; Saupe et al. 2018; Qiao et al. 2019). The divergence between niches over time may result from human-caused extirpation and extinctions (Dirzo et al. 2014; Johnson et al. 2017; Rutrough et al. 2019; Lim et al. 2022); lack of analogous environments (Svenning et al. 2011; Guevara 2019; Qiao et al. 2019); and/or undersampling, especially for rare species (Qiao et al. 2017). As such, model transferability is particularly tenuous when attempting to project (i.e., hindcast or forecast) to environments that are outside the scope of the training dataset (Varela et al. 2011; Guevara 2019). This challenge was illustrated by Maguire et al. (2016), who compared ENM approaches over the last 21,000 years by training models on occurrences from one time interval, projecting the models to another period, and then using empirical data from the projected period to evaluate model performance. They found that ENMs and paleoENMs performed poorly when projected to climatically dissimilar or temporally distant intervals compared with the one in which they were trained, and the authors noted this is particularly relevant to forecasting future distributions from purely contemporary data as novel climates emerge.

To increase the robustness of projections (particularly forecasts), there have been increasing calls to incorporate the fossil record into niche models, so as to expand the range and combination of environmental conditions experienced by organisms (Dietl and Flessa 2011; Varela et al. 2011; Dietl et al. 2015; Maguire et al. 2015). For example, Lima-Ribeiro et al. (2017) incorporated fossil data of jaguars (Panthera onca) into their niche model, which showed that abiotic tolerances for this species were broader than contemporary occurrences alone would suggest. Nógues-Bravo (2009) called for "pooling" species occurrences across multiple past time periods (Fig. 1D) as a way to better approximate species' fundamental niches, and several studies have illustrated that pooled niches outperform niches derived from a single time or place when projecting to different environments (Broennimann and Guisan 2008; Maiorano et al. 2013; Metcalf et al. 2014). To some extent, the time-averaged nature of the fossil record provides a greater chance of detecting occurrences over century to millennial timescales, and thus provides a built-in mechanism for "pooling" occurrence data (Behrensmeyer et al. 2000; Kidwell 2002, 2013; Patzkowsky 2017). Time averaging, however, may also alter the signal of corresponding environments or lead to occurrence-environment mismatches (Behrensmeyer et al. 2000).

Overall, the use of fossils requires careful evaluation to determine whether the environments a model is trained on are comparable to environments in the projection time period, to avoid the same pitfalls encountered by those training solely on contemporary occurrences. Metrics such as multivariate environmental similarity surfaces (Elith et al. 2010), extrapolation detection (Mesgaran et al. 2014), or mobility-oriented parity analysis (Owens et al. 2013) can be used to assess environmental similarity across space or time. Some studies have also used clamping, whereby values that occur outside training bounds are instead assigned the value of their nearest environmental space to reduce extrapolation or are simply set to have a suitability score of zero. These analytics can help identify when models are extrapolating beyond their training data, or prevent extrapolation altogether, but few papers have examined these issues using the fossil record to date.

Using ENMs to Understand Paleobiogeographic Patterns and Processes

Although there are still obstacles to overcome, methodological advances over the last decade have started to address taphonomic biases and sample size issues and have increased the utility of the fossil record for modeling ecological niches. In turn, paleoENMs have contributed greatly to our understanding of ENMs through studies that have examined niche change and the limits of model transferability. This progress should widen the applicability of paleoENMs to paleobiogeography. Even though the proportion of niche-modeling studies relying on the fossil record has remained relatively stable over the last decade (Fig. 2), ENMs and paleoENMs are now being applied to a wide variety of past time periods across a range of taxa (Figs. 3-5). We thus turn to the contributions of ENMs to paleobiogeography, grouping our discussion into five broad (and often overlapping) categories: (1) patterns and associated drivers of range dynamics; (2) phylogeography and withinlineage dynamics; (3) macroevolutionary patterns and processes, including niche change, speciation, and extinction; (4) drivers of biodiversity assembly; and (5) conservation paleobiogeography. We have not attempted to provide an exhaustive overview, but rather focus on highlighting recently published papers that use ENMs or paleoENMs to explore aspects of paleobiogeography. Overall, despite the relative rarity of niche modeling in studies that involve the fossil record, we demonstrate that ENMs and paleoENMs provide powerful insight into paleobiogeographic patterns and processes.

Patterns and Associated Drivers of Range Dynamics

Occurrence data allow us to infer potential geographic distributions of species and, in some cases, determine patterns of range shifts through time. For example, occurrence data are often used to infer the contemporary ranges of many species across Earth with relatively high confidence (Rondinini et al. 2006; Fourcade 2016; Merow et al. 2017). Utilizing spatiotemporally explicit contemporary data has also enabled the detection of range shifts that occurred within the most recent decades to centuries, making it apparent that the ranges of many species are shifting in response to recent anthropogenic climate change and landscape transformation (Parmesan and Yohe 2003; Moritz et al. 2008; Pecl et al. 2017; Lenoir et al. 2020). Similarly, fossil observations have provided compelling evidence of changes in species' distributions across millennial timescales (Graham et al. 1996; Lyons 2003; Precht and Aronson 2004; Giesecke et al. 2017), providing the foundation for understanding the impacts of species-specific distributional changes on communities (e.g., Williams et al. 2004) or on evolutionary change (e.g., Davis and Shaw 2001; Alsos et al. 2009).

Even for the most densely sampled species, however, a variety of biases may affect the inference of species ranges from observational data alone, and these gaps become more pronounced when inferring range changes through time. By estimating the portion of a species' potential range, where occurrences or fossils are not observed, ENMs can provide a more complete picture of both contemporary and past distributions. ENMs are thus a useful tool to understand species' range expansions, contractions, and shifts (e.g., Stigall Rode and Lieberman 2005; Maguire and Stigall 2009; Rindel et al. 2021; Wendt et al. 2022) and to formulate hypotheses about ecological dynamics. Finally, paleoENMs can help illuminate whether the absence of fossil occurrences in an area is likely to reflect true absences attributable to aspects of species' ecology or evolution, or stems from taphonomic biases (e.g., Inman et al. 2021) (see "Taphonomic Bias" section).

Two recent papers illustrate the power of using multiple approaches and lines of evidence, including paleoENMs, to help interpret patterns of range change through time. Wendt et al. (2022) inferred North American bison (Bison spp.) distributions for twenty-two 1,000 year time slices since the LGM. The authors examined species' range dynamics by comparing the paleoENMderived estimates of range shifts with occupancy-based estimates of abundance change. They also combined information from multiple time slices to determine which variables to retain in the paleoENMs, enabling evaluation of the importance of climatic drivers such as thermal stress and aridity in structuring range dynamics. Similarly, Rindel et al. (2021) modeled the distribution of guanacos (Lama guanicoe) in southern South America across four time periods during the late Quaternary, finding that, contrary to expectations, this species' geographic distribution was not contiguous in the past and had decreased substantially through time, despite strong demographic growth. Using contrasts between the modeled distributions and other independent proxy data from zooarchaeological sites, Rindel et al. (2021) inferred that patterns of human subsistence on guanacos were strongly reflective of guanaco distribution; that is, early humans in the region preyed on guanacos where they were available, but switched to exploit other species in areas of low suitability for guanacos. Overall, these two example studies illustrate how paleoENM can complement other paleo-data and analytical approaches to provide details about the temporal dynamics of geographic ranges and deepen inferences about the drivers of past spatiotemporal patterns.

Phylogeography and Within-Lineage Dynamics

One of the most frequent applications of ENMs in our literature review is to understand the phylogeographic structure of contemporary species. These studies implicitly examine range shifts through time, although the goal of applying ENMs is typically not to understand range shift dynamics per se, but rather to explain contemporary population genetic structure. To this end, hindcast ENMs or paleoENMs are used to corroborate inferences gained from molecular evidence, such as the location of past climate refugia, species' demographic changes through time, and/or hypotheses of vicariance, allopatry, or hybridization within species or between extant sister species (e.g., Alvarado-Serrano and Knowles 2014; Gavin et al. 2014; Sawyer et al. 2019; Rico et al. 2021; Amat and Escoriza 2022).

Despite clear links to past species' distributions, the majority of ENM studies that have made projections to the past do not rely on fossil evidence, paleoENM or otherwise (Fig. 2). Instead, authors primarily hindcast ENMs calibrated using contemporary occurrence data alone (Fig. 5), with varying degrees of integration between the molecular approaches and niche modeling (e.g., Gavin et al. 2014; Wieringa et al. 2020). Furthermore, the majority of studies examine patterns during the Quaternary only, typically hindcasting to the mid-Holocene, the LGM, and more recently,

the last interglacial period (Figs. 4, 5), likely due to the widespread availability of environmental predictor data for these time intervals. In rare cases, the hindcasting results are compared with fossil locality data (e.g., Iannella et al. 2017; Li et al. 2022), but this is not typical. As Gavin et al. (2014: p. 43) noted while calling for tighter integration of fossils, genetics, and niche models, hindcasting contemporary niche models "is cheaper and easier than those [approaches] that rely strictly on fossil or genetic data for inference."

Despite the overall paucity of studies, a notable subset of papers relies more substantially on fossil evidence in tandem with phylogeographic data to explore hypotheses about phylogenetic history, through either the explicit integration of paleoENMs or by statistically validating hindcast ENMs using the fossil record. For example, Lagerholm et al. (2017) used both hindcast ENMs and paleoENMs, coupled with data and demographic modeling from ancient DNA, to examine population change for two sister species of ptarmigan (Lagopus lagopus and Lagopus muta). Lagerholm et al. (2017) ultimately relied on estimates of past environmental suitability resulting from paleoENMs, in part because the models trained on fossil data performed better than the hindcast niche models. In another example, Napier et al. (2019) used paleoENMs built with fossil pollen data to explain the biogeography of three Alaskan species of alders (Alnus) that resulted from vicariance in three different glacial refugia followed by postglacial dispersal and coalescence. Other authors have used paleoENMs to locate hybridization events (Rocha et al. 2022) or compare the niches of sister species (Feng et al. 2017; Melchionna et al. 2018) (see also "Niche Change" section). In some cases, climatic niche differences directly maintain species boundaries by limiting gene flow (De La Torre et al. 2014; Litvinchuk et al. 2020).

More recently, ENMs have been used to parameterize demographic simulations that explore hypotheses about spatiotemporal patterns of population demography (e.g., Brown and Knowles 2012). For example, Prates et al. (2016) translated hindcast niche models into several related parameters: initial ancestral areas, friction surfaces that indicate the difficulty of species' movement across the landscape, and overall carrying capacity. These data were then integrated with demographic simulations to estimate genetic diversity, finding that species' responses to climate shifts were determined by their dispersal abilities. Likewise, Metcalf et al. (2014) integrated fossil data, genetic data, paleoENMs, and demographic modeling to examine Bison population and range dynamics through time. Here, fossil data from individual time slices and pooled data over the entire late Quaternary were used to generate paleoENMs and predict distributions at different temporal steps (42, 30, 21, and 6 ka, and preindustrial), which then guided the creation of alternate demographic models of population history.

A suite of studies have extended this hypothesis-testing approach by combining fossil occurrences, genetic data, and ENMs with spatially explicit population models and pattern-oriented modeling (Canteri et al. 2022; Fordham et al. 2022, Pilowsky et al. 2022b). Here, paleoENMs are first used to generate the *n*dimensional hypervolume of climate suitability (e.g., Fig. 1). Subsamples are then drawn from that hypervolume to generate many different bioclimatic envelope models. These subsampled niches can be coupled with stochastic population models and other input data to simulate spatially explicit population dynamics. The fossilcalibrated climate suitability can also be used to estimate some of the constraining parameters such as maximum abundance (Fordham et al. 2021, 2022). This framework has been used to explore patterns of population change, extirpation, and extinction, providing detailed insights into the ecological processes underlying demographic change (Pilowsky et al. 2022a) (see also "Extinction Causes" section).

Overall, several studies using paleoENMs have demonstrated the power of using niche models combined with other data and approaches in an explicit hypothesis-testing approach. In all cases, the use of fossils improves the resolution of niche estimates and extends the temporal range of investigation, allowing the evaluation of more detailed biogeographic hypotheses.

Macroevolutionary Patterns and Processes

Niche Change. Most of the approaches in the preceding sections estimate demographic change and within-lineage diversification under the assumption that niches are static within lineages through time. However, across longer time spans, this assumption is increasingly likely to be inaccurate. The existence of species in almost every environment on Earth is evidence of significant niche evolution over the history of life, but when, and at what rates, niches evolve remains widely debated. Determining the dynamics of niche evolution is a fundamental biological question that can help to elucidate evolutionary and ecological processes, including geographic modes of speciation and extinction (Graham et al. 2004; Wiens et al. 2010; Quintero et al. 2022) and persistent patterns such as latitudinal diversity gradients (Diniz-Filho et al. 2007; Pyron and Burbrink 2009; Romdal et al. 2013).

Identifying true instances of niche evolution can be difficult, however, and so throughout this section we refer to "niche change" rather than "niche evolution." Most correlative modeling approaches estimate the realized niche (Saupe et al. 2012). Changes in the realized niche do not necessarily correspond to changes in the fundamental niche, and apparent niche shifts may instead reflect dispersal events into new habitats, changes in biotic interactions that broaden or narrow the range of environments available to a species, or environmental changes that influence the availability of suitable conditions independent of species' interactions (among other factors, including adequate sampling or changes in preservation biases through time). Consequently, rates of niche change can often be overestimated (Saupe et al. 2018; Owens et al. 2020), and niche comparisons must be conditioned on the environments existing and accessible to species at any given time.

Niche modeling is a useful tool that can help to constrain the tempo and mode of niche change. ENMs and paleoENMs can be used to estimate the rate and relative frequency of niche change across speciation events within clades (Peterson et al. 1999; Wiens and Graham 2005; Knouft et al. 2006; Losos 2008; Evans et al. 2009; Vieites et al. 2009; Nyári and Reddy 2013) or within individual, evolving lineages (Martínez-Meyer and Peterson 2006; Stigall 2012, 2014; Saupe et al. 2014). Niche change across clades can be quantified using extant species only (but see Meseguer et al. 2015; Lawing et al. 2016; Rolland et al. 2018; Jezkova 2020; Rivera et al. 2020; Zhang et al. 2022), but determining rates of niche change within individual, evolving lineages requires the temporal perspective provided by fossil data (Svenning et al. 2011; Stigall 2012; Fritz et al. 2013).

PaleoENM studies of within-lineage niche change typically quantify species' niches using fossil data from multiple temporal snapshots, which are then compared through time using measures of (dis)similarity. Analyses are often performed at the species level, although higher taxonomic units are used more frequently deeper in time. Most within-lineage studies have focused on marine plankton (Antell et al. 2021), marine invertebrates (Stigall 2012, 2014; Hopkins et al. 2014; Saupe et al. 2014; Patzkowsky and Holland 2016), or terrestrial pollen (Martínez-Meyer and Peterson 2006; Wang et al. 2023), because the fossil records for these groups have relatively fine spatiotemporal resolutions. Within-lineage niche dynamics are typically quantified over tens of thousands to millions of years, with examples from both the Quaternary (Antell et al. 2021; Wang et al. 2023) and pre-Quaternary (Malizia and Stigall 2011; Brame and Stigall 2014; Saupe et al. 2014; Patzkowsky and Holland 2016; Brisson et al. 2023) time periods. To date, withinlineage studies-regardless of temporal duration or time intervalhave recovered evidence for niche stability, even in the face of environmental change (Dudei and Stigall 2010; Saupe et al. 2014; Stigall 2014; Antell et al. 2021; Brisson et al. 2023). When niches were found to differ, these changes were often associated with biotic perturbations such as invasion events (Stigall 2012; Patzkowsky and Holland 2016) or massive biodiversity losses (Hopkins et al. 2014) and did not represent true evolutionary change but rather constriction of the previously occupied niche.

Similar to within-lineage analyses, across-lineage analyses have largely found support for niche conservatism. Across-lineage niche dynamics can be assessed by incorporating fossil information into phylogenetic comparative analyses (De La Torre et al. 2014; Meseguer et al. 2015; Lawing et al. 2016; Rolland et al. 2018; Rivera et al. 2020; Zhang et al. 2022). Niche stability and habitat tracking appear to be predominant over timescales of 10^5 – 10^7 years, while niche change may occur only occasionally and in response to significant environmental perturbations (Maguire and Stigall 2009; Carrier 2018; Jezkova 2020; Rivera et al. 2020; Sanz-Arnal et al. 2022; Chiarenza et al. 2023). Quantitative estimates of rates of niche change during diversification sensu Fritz et al. (2013) and Owens et al. (2020) have been attempted in only a handful of paleoENM studies, such as Rivera et al. (2020).

Niche dynamics can also be examined by estimating tolerances of extant taxa and assessing how well these models predict past distributions (Waterson et al. 2016; Di Febbraro et al. 2017; Saupe et al. 2019a; Brown et al. 2021). These approaches typically rely on hindcast ENMs trained on contemporary occurrences, rather than building paleoENMs at multiple temporal snapshots (but see Waterson et al. 2016; Sanz-Arnal et al. 2022). Analyses tend to characterize tolerances at the clade level to make predictions for stem lineages (Waterson et al. 2016; Saupe et al. 2019a). The success of these models in projecting suitable conditions in regions where lineages lived millions of years ago suggests conservatism in at least broadscale temperature and precipitation tolerances across cladogenic events. For example, ENMs for Southern Hemisphere bird clades today predict fossil distributional data for ancestors living 50 Ma in the Northern Hemisphere (Saupe et al. 2019a).

Although niche stability has been found both within lineages and across speciation events over a range of timescales, patterns of niche change have been reported for some clades, times, and regions (e.g., Malizia and Stigall 2011; Waterson et al. 2016; Di Febbraro et al. 2017; Jezkova 2020; Brown et al. 2021; Wang et al. 2023). For example, around 25% of the studied plant taxa over the last 18,000 years exhibited within-lineage niche lability, rather than stability (Wang et al. 2023). Similarly, on longer, million year timescales, support has been found for changed tolerances for at least some groups (Meseguer et al. 2015; Waterson et al. 2016). Endotherms, for example, may have greater lability in temperature tolerances than ectotherms (Rolland et al. 2018). Niches may be conserved over shorter temporal intervals and more labile over longer time spans (Pearman et al. 2008; Wiens et al. 2010; Peterson 2011). Overall, the fossil record can help elucidate the tempo and mode of niche dynamics when coupled with niche models. Identifying when niches change is important for determining the drivers of that change over time, and quantifying rates of change is critical for accurate projections of species' responses to anthropogenic climate change and associated conservation efforts. Determining when observed niche change is representative of true evolution toward novel tolerances, however, remains a key challenge in correlative modeling studies.

Speciation. PaleoENM-based methods can be used to identify potential drivers of speciation (Myers et al. 2015; Stigall 2015) by establishing whether speciation was associated with shifts in the environmental conditions occupied by species within clades. If new species occupy new niches after speciation or radiation events, this may indicate that environmental perturbations allowed access to new niche space, spurring diversification (Purcell and Stigall 2021). However, any such analysis must contextualize the perceived niche shifts on the occupancy and availability of the environmental background, and how these conditions have changed through time, taking taphonomic factors into account.

Speciation is often considered to occur allopatrically under an assumption of niche conservatism (Wiens et al. 2010). Increased diversification may therefore be expected to coincide with increased fragmentation of suitable abiotic conditions; the incorporation of ENMs into deep-time evolutionary studies can geographically and ecologically constrain the context of such diversification (Lawing et al. 2016; Saupe et al. 2019b). For example, studies have found that both Miocene equids and Ordovician brachiopods experienced higher speciation rates with minimal niche shift during intervals with lower connectivity of suitable conditions (Maguire and Stigall 2009; Purcell and Stigall 2021), presumably due to vicariance. Similarly, when climatic or tectonic changes produce new, but only marginally suitable, conditions, allopatric speciation with substantial niche diversification can result (Rivera et al. 2020). However, geographic barriers that prevent dispersal into newly suitable areas as well as high habitat connectivity may limit opportunities for allopatry and lead to reduced speciation (Stigall Rode and Lieberman 2005; Meseguer et al. 2015; Purcell and Stigall 2021). One major result that has emerged is that niche breadth affects responses to habitat fragmentation: generalists may diversify by vicariant speciation, while specialists are more likely to go extinct (Stigall 2015; Qiao et al. 2016; Rolland and Salamin 2016). Assemblagebased stratigraphic analyses of species' niche occupation (e.g., Brisson et al. 2023; Forsythe and Stigall 2023) draw similar conclusions to paleoENM studies in the same systems (Stigall 2023).

Extinction Causes. Habitat loss is frequently hypothesized as a major driver of both single-species extinction events and mass extinction events (e.g., Lima-Ribeiro et al. 2014; Reddin et al. 2022; Payne et al. 2023). PaleoENMs are useful tools with which to evaluate these hypotheses. Several paleoENM studies have investigated whether the loss, or rate of loss, of suitable conditions explains patterns of extinction or extirpation in a species or ecological group. Many of these studies have focused on the role of climate change in the late Quaternary megafaunal extinction, often comparing climate to human influence. For example, Di Febbraro et al. (2017) found that Eurasian megafauna with higher affinity for cold/arid habitat were more likely to go extinct in the late Pleistocene. Loss of favored environmental conditions was implicated in both local (Wang et al. 2018) and regional (Wang et al. 2021) extirpation of the woolly mammoth (Mammuthus primigenius) in North America, but human occupation was required to explain

the timing of woolly mammoth extinction in Eurasian (Fordham et al. 2022) and two South American proboscidean taxa (Lima-Ribeiro et al. 2013). The results of Fordham et al. (2022) corroborate findings from studies on several other Pleistocene mammals, which had suitable conditions available both before and after their respective extinction intervals (Varela et al. 2010; Elton and O'Regan 2014; Villavicencio et al. 2019). Models differ, however, in their approaches and scale, which makes direct comparison among studies difficult.

PaleoENM studies focused on earlier time intervals have also found evidence for environmental controls on extinction patterns. Carrier (2018) used sedimentary proxies for environment to confirm marine anoxia as the most likely cause of mollusk extinctions during the Cenomanian/Turonian boundary event. Suitable area and biodiversity both decreased over the Cenozoic in multiple tropical palm and mangrove subfamilies (Lim et al. 2022) and tropical podocarps (Robin-Champigneul et al. 2023). However, suitable conditions, even if reduced in spatial extent, were still available during extinction intervals for tropical palm and mangrove subfamilies (Lim et al. 2022), tropical podocarps (Robin-Champigneul et al. 2023), late Quaternary mammals (Varela et al. 2010; Elton and O'Regan 2014; Villavicencio et al. 2019; Fordham et al. 2022), and Late Cretaceous dinosaur families (Chiarenza et al. 2019), highlighting the unclear relationship between ENM-based habitable range estimates and biodiversity. Several studies have also used climate niche breadth to estimate extinction risk for a range of marine species under different climate change scenarios, including mollusks in the mid-Pliocene warm period (Saupe et al. 2015), shallow-marine bivalves from the Pliocene to modern (Collins et al. 2018), and cold water-specialized marine benthic invertebrates during hyperthermal events throughout the Phanerozoic (Reddin et al. 2020).

Overall, despite the applicability of paleoENM to macroevolutionary questions, such studies remain relatively rare. For example, there appears to be a relative paucity of ENM-based approaches examining pre-Quaternary extinctions, which perhaps reflects limited environmental data availability and/or uncertainty in projecting niches back onto reconstructed paleogeography during earlier time periods. Luckily, the availability of GCM-derived environmental predictor data for deep-time studies is improving (see "Environmental Reconstruction" section), and increased access to higher-resolution paleoclimate proxies and reconstructions may make paleoENMs more feasible for deeper-time studies, especially on regional scales.

Drivers of Biodiversity Assembly

Identifying the key factors behind the assembly and maintenance of biodiversity represents a central challenge within the fields of ecology and evolutionary biology and is particularly important in light of ongoing global change. The fossil record, covering extended time periods far beyond the reach of modern observations, offers an exciting avenue for exploring the dynamics of biodiversity assembly, especially in the context of long-term environmental change. Many studies have explored patterns and processes important to biodiversity assembly, demonstrating strong deterministic forces at work in structuring assemblages over millennia (McGill et al. 2005), alongside strong temporal (Graham et al. 1996; Lyons et al. 2016; Tóth et al. 2019) and spatial (Knight et al. 2020; Sundaram and Leslie 2021) variation in mechanisms governing assembly of both plant and animal diversity on timescales ranging from millennia to 300 Myr.

To date, however, these studies have seen limited reliance on paleoENMs and, despite their significance, paint an incomplete picture of the processes responsible for biotic assembly because they often lack extensive spatial coverage, sufficiently fine-scale spatiotemporal resolution, and detailed information on species co-occurrence. The demonstrated strong nonstationarity of past assembly processes (Graham et al. 1996; Lyons et al. 2016; Tóth et al. 2019; Knight et al. 2020; Sundaram and Leslie 2021), coupled with similar findings for the modern (Machac et al. 2011; Lamanna et al. 2014; Jarzyna et al. 2021), emphasizes the need for approaches that provide spatially and temporally explicit information on the co-occurrence of multiple species in the past. ENMs have proven a useful tool to obtain such data and enable comprehensive spatiotemporal explorations. Hindcast ENMs have been used to investigate the effects of past climatic changes on the composition of floristic assemblages (Wilson et al. 2021), avian assemblage structure in the Himalayas (Dong et al. 2021), carnivoran assemblages in the Americas (Arias-Alzate et al. 2020), and the impacts of climate and historical contingencies on beetle and mammal assemblage composition in areas of endemism in Mexico (Pinilla-Buitrago et al. 2018). To disentangle the processes of biodiversity assembly, however, researchers cannot rely solely on hindcasting ENMs. This is because ENMs operate under the assumption that species remain in equilibrium with their climatic niches, both in the modern and in the past, and that the relationships between species and their environments are temporally stationary-neither of which is likely to hold true over extended time spans (Graham et al. 1996; Lyons et al. 2016; Rowan et al. 2016; Tóth et al. 2019). Furthermore, with increasing temporal distance, model transferability becomes increasingly problematic (see "Model Transferability" section).

Initial efforts to utilize paleoENMs for understanding assembly processes in the past are underway. For example, Carotenuto et al. (2016) employed joint species distribution modeling based on fossil occurrences to quantify significant correlations between pairs of large mammal species occurring in the Eurasian fossil record for the last interglacial, the LGM, and the Holocene. Using these paleoENMs, the authors assessed the relative importance of two primary drivers of assembly of the mammalian biota: climatic conditions and ecological interactions. Their findings provide support for the increasing influence of climate filtering on the late Quaternary mammal fauna. Specifically, the co-occurrence of species was increasingly attributed to climatic variables and decreasingly to ecological interactions from the last interglacial period to the Holocene, corroborating prior findings (Lyons et al. 2016; Tóth et al. 2019).

Although in-depth investigations of this nature remain infrequent, increasing availability of multispecies occurrence data through databases such as the Neotoma Paleoecology Database (www.neotomadb.org; Williams et al. 2018) or the PBDB (https:// paleobiodb.org) should help facilitate use of paleoENMs for understanding past assembly processes. Although challenges in technical implementation still exist, the growing utilization of such resources underscores the significance of examining the spatial dimension alongside the temporal one, marking an encouraging shift that complements the inherent temporal strengths of the fossil record. We advocate that, moving forward, the fossil record should be a primary data source for modeling ecological niches and disentangling assembly processes when possible, with paleoENMs serving as the principal means for achieving this objective.

Conservation Paleobiogeography

One increasingly important use of paleoENMs is to inform aspects of biodiversity conservation using information from the past—a

subset of the emerging "conservation paleobiology" field (McGuire and Davis 2014). Many studies that couple genetic data with hindcast ENMs also forecast ENMs to the future to predict where species' refugia may occur or to understand whether species will be able to disperse to projected future suitable regions. Only a small subset of these studies, however, parameterize future projections using fossils (e.g., Lawing and Polly 2011), but those that do often find expanded suitable areas in the future, suggesting that inclusion of fossil data potentially ameliorates estimates of range loss by more fully approximating the niche of a species (e.g., Nogués-Bravo et al. 2016; Ivory et al. 2019; Jarvie et al. 2021). However, future prospects for biodiversity may be species dependent. For example, using an approach that developed both contemporary-occurrence ENMs as well as paleoENMs, Ivory et al. (2019) found that contemporary

well as paleoENMs, Ivory et al. (2019) found that contemporary species' ranges are limited by land use, rather than climate, and thus future projections of range loss and habitat fragmentation remain substantial due to the combined impacts of climate and land use. A paper focused on economically important tree species found that niche-modeling approaches anchored in fossil data reliably predicted current distributions (Macias-Fauria and Willis 2013), but found that the models worked least well for heavily managed tree species (as well as rare species and those with discontinuous distributions), echoing the findings of Ivory et al. (2019) that human land use and species management decreases the predictive ability of projections based on purely climate-based relationships. A compilation of projections of future abundance change for 187 fossil pollen taxa found that, while most species were projected to gain suitable conditions in the future, pooled paleoENMs predicted larger changes in abundance for both expanding and contracting species (Nogués-Bravo et al. 2016), reinforcing the need for incorporating multiple time periods in niche models to more accurately capture the full range of spatiotemporal change.

Some studies have also utilized paleoENMs to identify sites suitable for species' reintroduction (Lentini et al. 2018) and have examined the technical aspects related to the ability of paleoENMs and fossil data to be used in conservation planning. For example, Guevara (2019) discussed the use of paleoENMs to design sampling strategies for natural history collections based on gaps in the existing data coverage. Williams et al. (2013) built paleoENMs using fossil pollen occurrences from the LGM and early deglaciation (21-15 ka) coupled with paleoclimate simulations, which were then forecast to the present to predict modern distributions. These data were then fed into a framework for selecting potential reserves, and the correlation between these predicted modern reserve rankings and actual modern reserve rankings was determined. This approach builds on early work by Graham (1988) that examined what fossil distributions say about "refuges" and how these ancient distributions may apply to conservation, although Graham (1988) did not approach this from a niche modeling framework. Overall, although the ENM-based approach that included fossils was better than a strategy based purely on paleo-abiotic factors, it still only performed moderately well, indicating the limits of transferability highlighted in the "Model Transferability" section.

Frontiers of Paleoecological Niche Modeling

Important and innovative work has been done in all the areas we have covered in this review, but gaps remain in several domains. There is a clear need for further integration of fossil data in ENMs to address questions of distributional changes, niche dynamics, speciation, extinction, conservation, and method development. There are several areas that would benefit from in-depth follow-ups or fresh horizon scans. For example, Varela et al. (2011) highlighted challenges with environmental predictors and temporal uncertainty, and our SQLR analysis revealed that both hindcast ENMs and paleoENMs are highly driven by the availability of environmental layers at particular times (see the three annotated times in Fig. 5). We have additionally identified three areas that may provide fruitful complementary data or approaches for paleoENM development and integration: alternative occurrence data, traits, and phylogenies.

Alternative Paleo-Occurrence Data

Fossils are a rich natural archive, offering an unparalleled glimpse into the past. However, fossil occurrences are limited by preservation, so alternative sources of information on past distributions may help to fill in the gaps, particularly for late Quaternary time intervals. While museum specimens are a common source of past distributional information (Meineke et al. 2018a,b; Smith et al. 2023), other historical records are often underutilized. For instance, early accounts by naturalists often include remarks of species' sightings that often do not make it into museum occurrence databases (Carpenter et al. 2023). Similarly, a wealth of data lies within old news articles, offering opportunities to reconstruct extirpations and extinctions, providing additional occurrence data and context around human-induced species declines (Lee and Perry 2019; Carpenter et al. 2023). As many of these historical documents predate the most profound anthropogenic impacts, they can provide insights into the ecological niches of organisms before the last century of globalization. Historical documents themselves, however, are subject to their own biases and have shorter temporal bounds, making them suitable alternatives for only certain taxonomic groups.

Recent studies have also set about exploring more unconventional sources of paleo-occurrence data. For instance, Gámez-Brunswick and Rojas-Soto (2020) found that including rock art occurrence records when creating ENMs of desert bighorn sheep (Ovis canadensis nelsoni) in the mid-Holocene and present-day improved predictive accuracy of their models. Similarly, oral histories of Indigenous peoples often tie species to their environments, provide insights into their ecological interactions, and include narratives around species declines, which together might illuminate the drivers behind niche shifts (e.g., Yeakel et al. 2014; Wehi et al. 2018; Whaanga et al. 2018). Another possible avenue is the "biogeography of place names," particularly Indigenous place names, which often codify taxon occurrences that have otherwise been forgotten (Cox et al. 2002; Moore 2002). Increasingly, sedimentary ancient DNA is providing highly-resolved information on species' presences (Williams et al. 2023), primarily for the late Quaternary but increasingly for earlier time periods (Dalén et al. 2023). Alternative occurrence sources such as these offer insight into species' distributions in environments where fossils may not readily preserve, helping us to overcome some of the limitations for reconstructing ecological niches and distributional shifts through recent time. Importantly, however, these forms of alternative occurrence data are heavily biased toward the Quaternary; fossils remain the primary record of the biological world for older time periods.

Traits and PaleoENMs

Utilizing information on species' functional and life-history traits in conjunction with ENMs has the potential to offer mechanistic insights into past biodiversity patterns and dynamics. Traits encompass morphological, physiological, behavioral, and lifehistory characteristics of organisms, collectively forming a description of an organism's ecological (Eltonian) niche. In contemporary contexts, augmenting ENMs with information on traits can improve predictive accuracy (Kearney and Porter 2009) and inform the construction of future assemblages (van Bodegom et al. 2014). By linking species' organismal traits to environmental gradients, we can also begin to understand which traits exhibit the strongest connections with environmental perturbances. Traits also hold promise as explanatory factors for understanding differences among species with regard to their range shifts with climate change (Beissinger and Riddell 2021), although identifying trait characteristics closely linked to range dynamics has so far been mostly unsuccessful in the modern (MacLean and Beissinger 2017; Beissinger and Riddell 2021). However, the paleontological record potentially offers greater opportunities to identify such traits due to its temporal scope, enabling the study of long-term range expansions and contractions. Finally, in the modern, species' traits are emerging as an important tool to understand the mechanistic underpinnings of biodiversity assembly, particularly across large spatial and temporal scales, where experimental manipulation is unfeasible. By investigating assemblage trait structure, we can begin to untangle the roles of ecological interactions from environmental constraints in structuring assemblages (Cavender-Bares et al. 2009; HilleRisLambers et al. 2012; Adler et al. 2013). Despite the potential of trait-based analysis and recent calls for increased focus on taxonfree metrics (Eronen et al. 2010; Barnosky et al. 2017), the use of traitbased metrics to discern assembly processes in the paleoecological record remains rare (but see Knight et al. 2020).

Phylogenies and PaleoENMs

Phylogenies have been used widely to estimate diversification and extinction rates (Scholl and Wiens 2016; Title and Rabosky 2019; Quintero et al. 2023) and to trace the evolution of ecological niches (Liu et al. 2020; Quintero et al. 2022). Repeated calls have been made for the inclusion of fossils in phylogenetic reconstructions (Wagner 1995; Quental and Marshall 2010; Morlon 2014), with the mathematical necessity demonstrated (Louca and Pennell 2020) and various methods devised (e.g., Finarelli and Flynn 2006; Slater and Harmon 2013). Importantly, it has been shown that estimates of niche evolutionary rates and ancestral-state reconstructions might be prone to errors if species' niches are incompletely or erroneously characterized (Saupe et al. 2018), as is often the case when only present-day occurrences are considered. A framework has recently been developed for the integration of fossil-informed phylogenies with paleoENMs for ancestral-range and ancestralstate reconstructions (Lawing 2021); Guillory and Brown (2021) even had some success in reconstructing simulated ancestral niches using only hindcast ENMs. The potential thus exists for paleoENMs to inform phylogenetic reconstructions and provide further insight into niche and other ecological and evolutionary dynamics across varying temporal scales.

Conclusions

We explored how ecological niche modeling has contributed to understanding the spatiotemporal distribution of past biodiversity and past ecological and evolutionary processes. Since the inception of niche-modeling tools and their first application to the fossil record, the number of studies using ecological niche models (ENMs) for understanding paleobiogeography has increased substantially and the proportion of studies that include evidence from the fossil record appear to have reached a steady state. ENMs now contribute considerably to understanding many different aspects of paleobiogeography, including questions centered on species' range dynamics; niche change; rates of speciation, extinction, and diversification; and conservation. While the vast majority of studies remain centered on the Quaternary, an increasing number of studies apply ENMs to pre-Quaternary times, broadening the utility of the approach. With increasing availability of fossil data via paleoecological and paleobiological databases, along with continued advances in reconstructing past environments at finer spatial and temporal resolutions, ongoing development of more statistically robust modeling techniques, and deeper integration with traits and phylogenies, paleoENMs are well positioned to substantially contribute to extending paleobiogeographic insights.

Acknowledgments. We thank the editors of the special issue for the invitation to contribute to this special issue, and M. Patzkowsky and C. Myers for comments that greatly improved the article. This work was supported by National Science Foundation (NSF) Division of Earth Sciences (EAR) 2149416 and EAR 1750597 to J.L.B., NSF EAR 2149419 to M.A.J., and Natural Environment Research Council NE/V011405/1 and a Leverhulme Prize to E.E.S.

Competing Interests. The authors declare no competing interests.

Data Availability Statement. The base data for the SQLR analysis are available at Dryad (https://doi.org/10.5061/dryad.m37pvmd9n). Supplementary Appendix 1 and a full reproducible script is available via Zenodo (https://doi.org/10.5281/zenodo.10905846).

Literature Cited

- Adler, P. B., A. Fajardo, A. R. Kleinhesselink, and N. J. B. Kraft. 2013. Traitbased tests of coexistence mechanisms. *Ecology Letters* 16:1294–1306.
- Alsos, I. G., T. Alm, S. Normand, and C. Brochmann. 2009. Past and future range shifts and loss of diversity in dwarf willow (*Salix herbacea L.*) inferred from genetics, fossils and modelling. *Global Ecology and Biogeography* 18:223–239.
- Alvarado-Serrano, D. F., and L. L. Knowles. 2014. Ecological niche models in phylogeographic studies: applications, advances and precautions. *Molecular Ecology Resources* 14:233–248.
- Amat, F., and D. Escoriza. 2022. Biogeographic inferences on the evolutionary history of the king cobra (*Ophiophagus hannah*, Cantor 1836) species complex. Zoological Studies 61:e28.
- Antell, G. T., I. S. Fenton, P. J. Valdes, and E. E. Saupe. 2021. Thermal niches of planktonic foraminifera are static throughout glacial–interglacial climate change. *Proceedings of the National Academy of Sciences USA* 118: e2017105118.
- Arias-Alzate, A., J. F. González-Maya, J. Arroyo-Cabrales, R. A. Medellín, and E. Martínez-Meyer. 2020. Environmental drivers and distribution patterns of carnivoran assemblages (Mammalia: Carnivora) in the Americas: past to present. *Journal of Mammalian Evolution* 27:759–774.
- Arruda, D. M., C. E. Schaefer, R. S. Fonseca, R. R. Solar, and E. I. Fernandes-Filho. 2018. Vegetation cover of Brazil in the last 21 ka: new insights into the Amazonian refugia and Pleistocenic arc hypotheses. *Global Ecology and Biogeography* 27:47–56.
- Baker, D. J., I. M. D. Maclean, M. Goodall, and K. J. Gaston. 2022. Correlations between spatial sampling biases and environmental niches affect species distribution models. *Global Ecology and Biogeography* 31:1038–1050.
- Barnosky, A. D., E. A. Hadly, P. Gonzalez, J. Head, P. D. Polly, A. M. Lawing, J. T. Eronen, et al. 2017. Merging paleobiology with conservation biology to guide the future of terrestrial ecosystems. *Science* 355:eaah4787.
- Behrensmeyer, A. K., S. Kidwell, and R. Gastaldo. 2000. Taphonomy and paleobiology. *Paleobiology* 26:103–147.

- Block, S., F. Saltré, M. Rodríguez-Rey, D. A. Fordham, I. Unkel, and C. J. Bradshaw. 2016. Where to dig for fossils: combining climate-envelope, taphonomy and discovery models. *PLoS ONE* 11:e0151090.
- Blois, J. L., P. L. Zarnetske, M. C. Fitzpatrick, and S. Finnegan. 2013. Climate change and the past, present, and future of biotic interactions. *Science* 341: 499–504.
- Bottjer, D. J., and D. Jablonski. 1988. Paleoenvironmental patterns in the evolution of post-Paleozoic benthic marine invertebrates. *Palaios* 3:540–560.
- Braconnot, P., M. Kageyama, S. P. Harrison, B. L. Otto-Bliesner, A. Abe-Ouchi, M. Willé, J.-Y. Peterschmitt, and N. Caud. 2021. PMIP key dates and achievements over the last 30 years. Past Global Changes Magazine 29:66–67.
- Brame, H.-M. R., and A. L. Stigall. 2014. Controls on niche stability in geologic time: congruent responses to biotic and abiotic environmental changes among Cincinnatian (Late Ordovician) marine invertebrates. *Paleobiology* 40:70–90.
- Brisson, S. K., J. Q. Pier, J. A. Beard, A. M. Fernandes, and A. M. Bush. 2023. Niche conservatism and ecological change during the Late Devonian mass extinction. *Proceedings of the Royal Society B* 290:20222524.
- Broennimann, O., and A. Guisan. 2008. Predicting current and future biological invasions: both native and invaded ranges matter. *Biology Letters* 4: 585–589.
- Broennimann, O., M. C. Fitzpatrick, P. B. Pearman, B. Petitpierre, L. Pellissier, N. G. Yoccoz, W. Thuiller, M. Fortin, C. Randin, and N. E. Zimmermann. 2012. Measuring ecological niche overlap from occurrence and spatial environmental data. *Global Ecology and Biogeography* 21:481–497.
- Brooke, C. F., C. W. Marean, C. D. Wren, H. Fritz, and J. A. Venter. 2022. Using functional groups to predict the spatial distribution of large herbivores on the Palaeo-Agulhas Plain, South Africa, during the Last Glacial Maximum. *Journal of Quaternary Science* 37:1056–1068.
- Brown, J. L., and L. L. Knowles. 2012. Spatially explicit models of dynamic histories: examination of the genetic consequences of Pleistocene glaciation and recent climate change on the American pika. *Molecular Ecology* 21: 3757–3775.
- Brown, J. L., D. J. Hill, A. M. Dolan, A. C. Carnaval, and A. M. Haywood. 2018. PaleoClim, high spatial resolution paleoclimate surfaces for global land areas. *Scientific Data* 5:180254.
- Brown, M. J. M., T. J. Brodribb, and G. J. Jordan. 2021. No-analogue associations in the fossil record of southern conifers reveal conservatism in precipitation, but not temperature axes. *Global Ecology and Biogeography* **30**:2455–2466.
- Canteri, E., S. C. Brown, N. M. Schmidt, R. Heller, D. Nogues-Bravo, and D. A. Fordham. 2022. Spatiotemporal influences of climate and humans on muskox range dynamics over multiple millennia. *Global Change Biology* 28: 6602–6617.
- Cardinale, B. J., J. E. Duffy, A. Gonzalez, D. U. Hooper, C. Perrings, P. Venail, A. Narwani, et al. 2012. Biodiversity loss and its impact on humanity. *Nature* 486:59–67.
- Carotenuto, F., M. Di Febbraro, M. Melchionna, S. Castiglione, F. Saggese, C. Serio, A. Mondanaro, F. Passaro, A. Loy, and P. Raia. 2016. The influence of climate on species distribution over time and space during the late Quaternary. *Quaternary Science Reviews* 149:188–199.
- Carpenter, J. K., G. L. Perry, and J. M. Wilmshurst. 2023. Palaeoecological and historical observations of an endemic New Zealand bird (*Strigops habroptila*, kākāpō) reveal shifting drivers of decline during 800 years of human settlement. *Global Ecology and Conservation* 43:e02433.
- Carrier, A. E. 2018. Investigation of molluscan survivorship across the Cenomanian–Turonian Boundary Event using ecological niche modeling. M.S. thesis, University of New Mexico, Albuquerque.
- Cavender-Bares, J., K. H. Kozak, P. V. A. Fine, and S. W. Kembel. 2009. The merging of community ecology and phylogenetic biology. *Ecology Letters* 12: 693–715.
- Chase, J. M., and M. A. Leibold. 2003. Ecological niches: linking classical and contemporary approaches. University of Chicago Press, Chicago.

- Chevalier, M., B. A. S. Davis, O. Heiri, H. Seppä, B. M. Chase, K. Gajewski, T. Lacourse, et al. 2020. Pollen-based climate reconstruction techniques for late Quaternary studies. *Earth-Science Reviews* 210:103384.
- Chiarenza, A. A., P. D. Mannion, D. J. Lunt, A. Farnsworth, L. A. Jones, S.-J. Kelland, and P. A. Allison. 2019. Ecological niche modelling does not support climatically-driven dinosaur diversity decline before the Cretaceous/Paleogene mass extinction. *Nature Communications* 10:1091.
- Chiarenza, A. A., A. M. Waterson, D. N. Schmidt, P. J. Valdes, C. Yesson, P. A. Holroyd, M. E. Collinson, et al. 2023. 100 million years of turtle paleoniche dynamics enable the prediction of latitudinal range shifts in a warming world. *Current Biology* 33:109–121.e3.
- Collins, K. S., S. M. Edie, G. Hunt, K. Roy, and D. Jablonski. 2018. Extinction risk in extant marine species integrating palaeontological and biodistributional data. *Proceedings of the Royal Society B* 285:20181698.
- Colwell, R., and T. Rangel. 2009. Hutchinson's duality: the once and future niche. Proceedings of the National Academy of Sciences USA 106:19651– 19658.
- Cox, J. J., D. S. Maehr, and J. L. Larkin. 2002. The biogeography of faunal place names in the United States. *Conservation Biology* 16:1143–1150.
- Dalén, L., P. D. Heintzman, J. D. Kapp, and B. Shapiro. 2023. Deep-time paleogenomics and the limits of DNA survival. *Science* 382:48–53.
- Davis, E. B., J. L. McGuire, and J. D. Orcutt. 2014. Ecological niche models of mammalian glacial refugia show consistent bias. *Ecography* 37: 1133–1138.
- Davis, M. B., and R. G. Shaw. 2001. Range shifts and adaptive responses to Quaternary climate change. *Science* 292:673–679.
- De La Torre, A. R., D. R. Roberts, and S. N. Aitken. 2014. Genome-wide admixture and ecological niche modelling reveal the maintenance of species boundaries despite long history of interspecific gene flow. *Molecular Ecology* 23:2046–2059.
- Dietl, G. P., and K. W. Flessa. 2011. Conservation paleobiology: putting the dead to work. *Trends in Ecology and Evolution* 26:30–37.
- Dietl, G. P., S. M. Kidwell, M. Brenner, D. A. Burney, K. W. Flessa, S. T. Jackson, and P. L. Koch. 2015. Conservation paleobiology: leveraging knowledge of the past to inform conservation and restoration. *Annual Review of Earth and Planetary Sciences* 43:79–103.
- Di Febbraro, M., F. Carotenuto, S. Castiglione, D. Russo, A. Loy, L. Maiorano, and P. Raia. 2017. Does the jack of all trades fare best? Survival and niche width in Late Pleistocene megafauna. *Journal of Biogeography* 44:2828–2838.
- Diniz-Filho, J. A. F., T. F. L. V. B. Rangel, L. M. Bini, and B. A. Hawkins. 2007. Macroevolutionary dynamics in environmental space and the latitudinal diversity gradient in New World birds. *Proceedings of the Royal Society B* 274:43–52.
- Dirzo, R., H. S. Young, M. Galetti, G. Ceballos, N. J. Isaac, and B. Collen. 2014. Defaunation in the Anthropocene. *Science* 345:401–406.
- Dong, F., C.-M. Hung, S.-H. Li, and X.-J. Yang. 2021. Potential Himalayan community turnover through the Late Pleistocene. *Climatic Change* 164:6.
- Dudei, N. L., and A. L. Stigall. 2010. Using ecological niche modeling to assess biogeographic and niche response of brachiopod species to the Richmondian Invasion (Late Ordovician) in the Cincinnati Arch. *Palaeogeography, Palaeoclimatology, Palaeoecology* 296:28–43.
- Elith, J., and J. R. Leathwick. 2009. Species distribution models: ecological explanation and prediction across space and time. *Annual Review of Ecology*, *Evolution, and Systematics* 40:677–697.
- Elith, J., M. Kearney, and S. Phillips. 2010. The art of modelling range-shifting species. *Methods in Ecology and Evolution* 1:330–342.
- Elton, S., and H. J. O'Regan. 2014. Macaques at the margins: the biogeography and extinction of *Macaca sylvanus* in Europe. *Quaternary Science Reviews* 96: 117–130.
- Eronen, J. T., P. D. Polly, M. Fred, J. Damuth, D. C. Frank, V. Mosbrugger, C. Scheidegger, N. Chr. Stenseth, and M. Fortelius. 2010. Ecometrics: the traits that bind the past and present together. *Integrative Zoology* 5:88–101.
- Evans, M. E. K., S. A. Smith, R. S. Flynn, and M. J. Donoghue. 2009. Climate, niche evolution, and diversification of the "bird-cage" evening primroses (*Oenothera*, Sections Anogra and Kleinia). American Naturalist 173:225–240.
- Fagoaga, A., H.-A. Blain, R. Marquina-Blasco, C. Laplana, N. Sillero, C. M. Hernández, C. Mallol, B. Galván, and F. J. Ruiz-Sánchez. 2019. Improving

the accuracy of small vertebrate-based palaeoclimatic reconstructions derived from the Mutual Ecogeographic Range. A case study using geographic information systems and UDA-ODA discrimination methodology. *Quaternary Science Reviews* **223**:105969.

- Farnsworth, A., D. J. Lunt, C. L. O'Brien, G. L. Foster, G. N. Inglis, P. Markwick, R. D. Pancost, and S. A. Robinson. 2019. Climate sensitivity on geological timescales controlled by nonlinear feedbacks and ocean circulation. *Geophysical Research Letters* 46:9880–9889.
- Feng, X., T. C. S. Anacleto, and M. Papeş. 2017. Climatic similarity of extant and extinct Dasypus armadillos. Journal of Mammalian Evolution 24:193–206.
- Finarelli, J. A., and J. J. Flynn. 2006. Ancestral state reconstruction of body size in the Caniformia (Carnivora, Mammalia): the effects of incorporating data from the fossil record. *Systematic Biology* **55**:301–313.
- Fordham, D. A., F. Saltre, S. Haythorne, T. M. L. Wigley, B. L. Otto-Bliesner, K. C. Chan, and B. W. Brook. 2017. PaleoView: a tool for generating continuous climate projections spanning the last 21 000 years at regional and global scales. *Ecography* 31:8.
- Fordham, D. A., S. Haythorne, S. C. Brown, J. C. Buettel, and B. W. Brook. 2021. poems: R package for simulating species' range dynamics using patternoriented validation. *Methods in Ecology and Evolution* 12:2364–2371.
- Fordham, D. A., S. C. Brown, H. R. Akçakaya, B. W. Brook, S. Haythorne, A. Manica, K. T. Shoemaker, et al. 2022. Process-explicit models reveal pathway to extinction for woolly mammoth using pattern-oriented validation. *Ecology Letters* 25:125–137.
- Forsythe, I. J., and A. L. Stigall. 2023. Insights for modern invasion ecology from biotic changes of the Clarksville Phase of the Richmondian Invasion (Ordovician, Katian). *Paleobiology* 49:493–508.
- Fourcade, Y. 2016. Comparing species distributions modelled from occurrence data and from expert-based range maps. Implication for predicting range shifts with climate change. *Ecological Informatics* 36:8–14.
- Franklin, J., A. J. Potts, E. C. Fisher, R. M. Cowling, and C. W. Marean. 2015. Paleodistribution modeling in archaeology and paleoanthropology. *Quaternary Science Reviews* 110:1–14.
- Fritz, S. A., J. Schnitzler, J. T. Eronen, C. Hof, K. Böhning-Gaese, and C. H. Graham. 2013. Diversity in time and space: wanted dead and alive. *Trends in Ecology and Evolution* 28:509–516.
- Gámez-Brunswick, C., and O. Rojas-Soto. 2020. New insights into palaeodistributions based on Holocene rock art. *Journal of Biogeography* 47:2543–2553.
- Gamisch, A. 2019. Oscillayers: a dataset for the study of climatic oscillations over Plio-Pleistocene time-scales at high spatial-temporal resolution. *Global Ecology and Biogeography* 28:1552–1560.
- Gavin, D. G., M. C. Fitzpatrick, P. F. Gugger, K. D. Heath, F. Rodríguez-Sánchez, S. Z. Dobrowski, A. Hampe, F. S. Hu, M. B. Ashcroft, and P. J. Bartlein. 2014. Climate refugia: joint inference from fossil records, species distribution models and phylogeography. *New Phytologist* 204:37–54.
- Giesecke, T., S. Brewer, W. Finsinger, M. Leydet, and R. H. W. Bradshaw. 2017. Patterns and dynamics of European vegetation change over the last 15,000 years. *Journal of Biogeography* **44**:1441–1456.
- Graham, C. H., S. R. Ron, J. C. Santos, C. J. Schneider, and C. Moritz. 2004. Integrating phylogenetics and environmental niche models to explore speciation mechanisms in dendrobatid frogs. *Evolution* 58:1781–1793.
- Graham, R. 1988. The role of climatic-change in the design of biological reserves —the paleoecological perspective for conservation biology. *Conservation Biology* 2:391–394.
- Graham, R., E. Lundelius, M. Graham, E. Schroeder, R. Toomey, E. Anderson, A. D. Barnosky, et al. 1996. Spatial response of mammals to late Quaternary environmental fluctuations. *Science* 272:1601–1606.
- Grinnell, J. 1917. The niche-relationships of the California thrasher. *Auk* 34: 427–433.
- Guevara, L. 2019. Paleodistribution modelling for planning the growth of natural history collections. *Revista Mexicana de Biodiversidad* 90. https:// doi.org/10.22201/ib.20078706e.2019.90.2953
- Guillory, W. X., and J. L. Brown. 2021. A new method for integrating ecological niche modeling with phylogenetics to estimate ancestral distributions. Systematic Biology 70:1033–1045.
- Guisan, A., and W. Thuiller. 2005. Predicting species distribution: offering more than simple habitat models. *Ecology Letters* 8:993–1009.

- Hadly, E. A., P. A. Spaeth, and C. Li. 2009. Niche conservatism above the species level. *Proceedings of the National Academy of Sciences USA* 106: 19707–19714.
- Hendricks, J. R., E. E. Saupe, C. E. Myers, E. J. Hermsen, and W. D. Allmon. 2014. The generification of the fossil record. *Paleobiology* 40:511–528.
- HilleRisLambers, J., P. B. Adler, W. S. Harpole, J. M. Levine, and M. M. Mayfield. 2012. Rethinking community assembly through the lens of coexistence theory. *Annual Review of Ecology, Evolution, and Systematics* 43:227–248.
- Holland, S. M., M. E. Patzkowsky, and K. M. Loughney. 2024. Stratigraphic paleobiology. *Paleobiology*. https://doi.org/10.1017/pab.2024.2.
- Holt, R. D. 2009. Bringing the Hutchinsonian niche into the 21st century: ecological and evolutionary perspectives. *Proceedings of the National Academy of Sciences USA* 106:19659–19665.
- Hope, A. G., E. Waltari, D. C. Payer, J. A. Cook, and S. L. Talbot. 2013. Future distribution of tundra refugia in northern Alaska. *Nature Climate Change* 3: 931–938.
- Hopkins, M. J., C. Simpson, and W. Kiessling. 2014. Differential niche dynamics among major marine invertebrate clades. *Ecology Letters* 17: 314–323.
- Hortal, J., A. Jiménez-Valverde, J. F. Gómez, J. M. Lobo, and A. Baselga. 2008. Historical bias in biodiversity inventories affects the observed environmental niche of the species. *Oikos* 117:847–858.
- Hutchinson, G. E. 1957. Concluding remarks. Cold Spring Harbor Symposium on Quantitative Biology 22:415–427.
- Iannella, M., F. Cerasoli, and M. Biondi. 2017. Unraveling climate influences on the distribution of the parapatric newts *Lissotriton vulgaris meridionalis* and *L. italicus. Frontiers in Zoology* 14:55.
- Inman, R., J. Franklin, T. Esque, and K. Nussear. 2018. Spatial sampling bias in the Neotoma paleoecological archives affects species paleo-distribution models. *Quaternary Science Reviews* 198:115–125.
- Inman, R., J. Franklin, T. Esque, and K. Nussear. 2021. Comparing sample bias correction methods for species distribution modeling using virtual species. *Ecosphere* 12:e03422.
- Ivory, S. J., J. Russell, R. Early, and D. F. Sax. 2019. Broader niches revealed by fossil data do not reduce estimates of range loss and fragmentation of African montane trees. *Global Ecology and Biogeography* 28:992–1003.
- Jablonski, D., S. Lidgard, and P. D. Taylor. 1997. Comparative ecology of bryozoan radiations: origin of novelties in cyclostomes and cheilostomes. *Palaios* 12:505–523.
- Jackson, S. T., and J. T. Overpeck. 2000. Responses of plant populations and communities to environmental changes of the late Quaternary. *Paleobiology* 26:194–220.
- Jackson, S. T., J. L. Betancourt, R. K. Booth, and S. T. Gray. 2009. Ecology and the ratchet of events: climate variability, niche dimensions, and species distributions. *Proceedings of the National Academy of Sciences USA* 106: 19685–19692.
- Jarvie, S., T. H. Worthy, F. Saltré, R. P. Scofield, P. J. Seddon, and A. Cree. 2021. Using Holocene fossils to model the future: distribution of climate suitability for tuatara, the last rhynchocephalian. *Journal of Biogeography* 48: 1489–1502.
- Jarzyna, M. A., I. Quintero, and W. Jetz. 2021. Global functional and phylogenetic structure of avian assemblages across elevation and latitude. *Ecology Letters* 24:196–207.
- Jezkova, T. 2020. Founder takes more: interspecific competition affects range expansion of North American mammals into deglaciated areas. *Journal of Biogeography* 47:712–720.
- Johnson, C. N., A. Balmford, B. W. Brook, J. C. Buettel, M. Galetti, L. Guangchun, and J. M. Wilmshurst. 2017. Biodiversity losses and conservation responses in the Anthropocene. *Science* 356:270–275.
- Karger, D. N., M. P. Nobis, S. Normand, C. H. Graham, and N. E. Zimmermann. 2023. CHELSA-TraCE21k—high-resolution (1 km) downscaled transient temperature and precipitation data since the Last Glacial Maximum. *Climate of the Past* 19:439–456.
- Kearney, M., and W. Porter. 2009. Mechanistic niche modelling: combining physiological and spatial data to predict species' ranges. *Ecology Letters* 12:334–350.
- Kidwell, S. M. 2002. Time-averaged molluscan death assemblages: palimpsests of richness, snapshots of abundance. *Geology* 30:803.

- Kidwell, S. M. 2013. Time-averaging and fidelity of modern death assemblages: building a taphonomic foundation for conservation palaeobiology. *Palaeontology* 56:487–522.
- Knight, C. A., J. L. Blois, B. Blonder, M. Macias-Fauria, A. Ordonez, and J.-C. Svenning. 2020. Community assembly and climate mismatch in Late Quaternary eastern North American pollen assemblages. *American Naturalist* 195:166–180.
- Knouft, J. H., J. B. Losos, R. E. Glor, and J. J. Kolbe. 2006. Phylogenetic analysis of the evolution of the niche in lizards of the *Anolis sagrei* group. *Ecology* 87:S29–S38.
- Kuemmerle, T., T. Hickler, J. Olofsson, G. Schurgers, and V. C. Radeloff. 2012. Reconstructing range dynamics and range fragmentation of European bison for the last 8000 years. *Diversity and Distributions* 18:47–59.
- Lagerholm, V. K., E. Sandoval-Castellanos, A. Vaniscotte, O. R. Potapova, T. Tomek, Z. M. Bochenski, P. Shepherd, et al. 2017. Range shifts or extinction? Ancient DNA and distribution modelling reveal past and future responses to climate warming in cold-adapted birds. *Global Change Biology* 23:1425–1435.
- Lamanna, C., B. Blonder, C. Violle, N. J. B. Kraft, B. Sandel, I. Šímová, J. C. Donoghue, et al. 2014. Functional trait space and the latitudinal diversity gradient. *Proceedings of the National Academy of Sciences USA* 111:13745–13750.
- Lawing, A. M. 2021. The geography of phylogenetic paleoecology: integrating data and methods to better understand biotic response to climate change. *Paleobiology* 47:178–197.
- Lawing, A. M., and P. D. Polly. 2011. Pleistocene climate, phylogeny, and climate envelope models: an integrative approach to better understand species' response to climate change. *PLoS ONE* 6:e28554.
- Lawing, A. M., P. D. Polly, D. K. Hews, and E. P. Martins. 2016. Including fossils in phylogenetic climate reconstructions: a deep time perspective on the climatic niche evolution and diversification of spiny lizards (*Sceloporus*). *American Naturalist* 188:133–148.
- Lee, F., and G. L. Perry. 2019. Assessing the role of off-take and source–sink dynamics in the extinction of the amphidromous New Zealand grayling (*Prototroctes oxyrhynchus*). Freshwater Biology 64:1747–1754.
- Lenoir, J., R. Bertrand, L. Comte, L. Bourgeaud, T. Hattab, J. Murienne, and G. Grenouillet. 2020. Species better track climate warming in the oceans than on land. *Nature Ecology and Evolution* 4:1044–1059.
- Lentini, P. E., I. A. Stirnemann, D. Stojanovic, T. H. Worthy, and J. A. Stein. 2018. Using fossil records to inform reintroduction of the kakapo as a refugee species. *Biological Conservation* 217:157–165.
- Li, W.-C., J. Huang, L.-L. Chen, R. A. Spicer, S.-F. Li, J. Liu, Y. Gao, et al. 2022. Podocarpium (Fabaceae) from the late Eocene of central Tibetan Plateau and its biogeographic implication. *Review of Palaeobotany and Palynology* 305: 104745.
- Lim, J. Y., H. Huang, A. Farnsworth, D. J. Lunt, W. J. Baker, R. J. Morley, W. D. Kissling, and C. Hoorn. 2022. The Cenozoic history of palms: global diversification, biogeography and the decline of megathermal forests. *Global Ecology and Biogeography* 31:425–439.
- Lima-Ribeiro, M. S., D. Nogués-Bravo, L. C. Terribile, P. Batra, and J. A. F. Diniz-Filho. 2013. Climate and humans set the place and time of Proboscidean extinction in late Quaternary of South America. *Palaeogeography*, *Palaeoclimatology*, *Palaeoecology* **392**:546–556.
- Lima-Ribeiro, M. S., J. Hortal, S. Varela, and J. A. F. Diniz-Filho. 2014. Constraint envelope analyses of macroecological patterns reveal climatic effects on Pleistocene mammal extinctions. *Quaternary Research* 82:260–269.
- Lima-Ribeiro, M. S., S. Varela, J. González-Hernández, G. de Oliveira, J. A. F. Diniz-Filho, and L. C. Terribile. 2015. EcoClimate: a database of climate data from multiple models for past, present, and future for macroecologists and biogeographers. *Biodiversity Informatics* 10. https://doi.org/10.17161/bi. v10i0.4955.
- Lima-Ribeiro, M. S., A. K. M. Moreno, L. C. Terribile, C. T. Caten, R. Loyola, T. F. Rangel, and J. A. F. Diniz-Filho. 2017. Fossil record improves biodiversity risk assessment under future climate change scenarios. *Diversity and Distributions* 23:922–933.
- Litvinchuk, S. N., N. A. Schepina, and A. Borzée. 2020. Reconstruction of past distribution for the Mongolian toad, *Strauchbufo raddei* (Anura: Bufonidae) using environmental modeling. *PeerJ* 8:e9216.
- Liu, H., Q. Ye, and J. J. Wiens. 2020. Climatic-niche evolution follows similar rules in plants and animals. *Nature Ecology and Evolution* 4:753–763.

- Losos, J. B. 2008. Phylogenetic niche conservatism, phylogenetic signal and the relationship between phylogenetic relatedness and ecological similarity among species. *Ecology Letters* 11:995–1003.
- Louca, S., and M. W. Pennell. 2020. Extant timetrees are consistent with a myriad of diversification histories. *Nature* 580:502–505.
- Lyons, S. K. 2003. A quantitative assessment of the range shifts of Pleistocene mammals. *Journal of Mammalogy* 84:385–402.
- Lyons, S. K., K. L. Amatangelo, A. K. Behrensmeyer, A. Bercovici, J. L. Blois, M. Davis, W. A. DiMichele, et al. 2016. Holocene shifts in the assembly of plant and animal communities implicate human impacts. *Nature* 529:80–83.
- Mace, G. M., B. Reyers, R. Alkemade, R. Biggs, F. S. Chapin, S. E. Cornell, S. Díaz, et al. 2014. Approaches to defining a planetary boundary for biodiversity. *Global Environmental Change* 28:289–297.
- Machac, A., M. Janda, R. R. Dunn, and N. J. Sanders. 2011. Elevational gradients in phylogenetic structure of ant communities reveal the interplay of biotic and abiotic constraints on diversity. *Ecography* **34**:364–371.
- Macias-Fauria, M., and K. J. Willis. 2013. Landscape planning for the future: using fossil records to independently validate bioclimatic envelope models for economically valuable tree species in Europe. *Global Ecology and Biogeography* 22:318–333.
- MacLean, S. A., and S. R. Beissinger. 2017. Species' traits as predictors of range shifts under contemporary climate change: a review and meta-analysis. *Global Change Biology* 23:4094–4105.
- Maguire, K. C., and A. L. Stigall. 2009. Using ecological niche modeling for quantitative biogeographic analysis: a case study of Miocene and Pliocene Equinae in the Great Plains. *Paleobiology* **35**:587–611.
- Maguire, K. C., D. Nieto Lugilde, M. C. Fitzpatrick, J. W. Williams, and J. L. Blois. 2015. Modeling species and community responses to past, present, and future episodes of climatic and ecological change. *Annual Review of Ecology*, *Evolution, and Systematics* 46:343–368.
- Maguire, K. C., D. Nieto-Lugilde, J. L. Blois, M. C. Fitzpatrick, J. W. Williams, S. Ferrier, and D. J. Lorenz. 2016. Controlled comparison of speciesand community-level models across novel climates and communities. *Proceedings of the Royal Society B* 283:20152817.
- Maiorano, L., R. Cheddadi, N. E. Zimmermann, L. Pellissier, B. Petitpierre, J. Pottier, H. Laborde, et al. 2013. Building the niche through time: using 13,000 years of data to predict the effects of climate change on three tree species in Europe. *Global Ecology and Biogeography* 22:302–317.
- Malizia, R. W., and A. L. Stigall. 2011. Niche stability in Late Ordovician articulated brachiopod species before, during, and after the Richmondian Invasion. *Palaeogeography, Palaeoclimatology, Palaeoecology* 311:154–170.
- Martínez-Meyer, E., and A. T. Peterson. 2006. Conservatism of ecological niche characteristics in North American plant species over the Pleistocene-to-Recent transition. *Journal of Biogeography* **33**:1779–1789.
- McGill, B. J., E. A. Hadly, and B. A. Maurer. 2005. Community inertia of Quaternary small mammal assemblages in North America. *Proceedings of the National Academy of Sciences USA* **102**:16701–16706.
- McGuire, J. L., and E. B. Davis. 2014. Conservation paleobiogeography: the past, present and future of species distributions. *Ecography* 37: 1092–1094.
- Meineke, E. K., T. J. Davies, B. H. Daru, and C. C. Davis. 2018a. Biological collections for understanding biodiversity in the Anthropocene. *Philosophical Transactions of the Royal Society B* 374:20170386.
- Meineke, E. K., C. C. Davis, and T. J. Davies. 2018b. The unrealized potential of herbaria for global change biology. *Ecological Monographs* 88:505–525.
- Melchionna, M., M. Di Febbraro, F. Carotenuto, L. Rook, A. Mondanaro, S. Castiglione, C. Serio, et al. 2018. Fragmentation of Neanderthals' preextinction distribution by climate change. *Palaeogeography, Palaeoclimatol*ogy, *Palaeoecology* 496:146–154.
- Merow, C., A. M. Wilson, and W. Jetz. 2017. Integrating occurrence data and expert maps for improved species range predictions. *Global Ecology and Biogeography* 26:243–258.
- Meseguer, A. S., J. M. Lobo, R. Ree, D. J. Beerling, and I. Sanmartín. 2015. Integrating fossils, phylogenies, and niche models into biogeography to reveal ancient evolutionary history: the case of *Hypericum* (Hypericaceae). *Systematic Biology* 64:215–232.

- Mesgaran, M. B., R. D. Cousens, and B. L. Webber. 2014. Here be dragons: a tool for quantifying novelty due to covariate range and correlation change when projecting species distribution models. *Diversity and Distributions* **20**: 1147–1159.
- Metcalf, J. L., S. Prost, D. Nogués-Bravo, E. G. DeChaine, C. Anderson, P. Batra, M. B. Araujo, A. Cooper, and R. P. Guralnick. 2014. Integrating multiple lines of evidence into historical biogeography hypothesis testing: a *Bison bison case study. Proceedings of the Royal Society B* 281:20132782–2625.
- Mithal, R., and D. G. Becker. 2006. The Janus database: providing worldwide access to ODP and IODP data. *Geological Society of London Special Publication* 267:253–259.
- **Moore, P. G.** 2002. Ravens (*Corvus corax corax* L.) in the British landscape: a thousand years of ecological biogeography in place-names. *Journal of Biogeography* **29**:1039–1054.
- Moreno-Amat, E., J. M. Rubiales, C. Morales-Molino, and I. García-Amorena. 2017. Incorporating plant fossil data into species distribution models is not straightforward: pitfalls and possible solutions. *Quaternary Science Reviews* 170:56–68.
- Moritz, C., J. L. Patton, C. J. Conroy, J. L. Parra, G. C. White, and S. R. Beissinger. 2008. Impact of a century of climate change on small-mammal communities in Yosemite National Park, USA. *Science* 322:261–264.
- Morlon, H. 2014. Phylogenetic approaches for studying diversification. *Ecology Letters* 17:508–525.
- Myers, C. E., A. L. Stigall, and B. S. Lieberman. 2015. PaleoENM: applying ecological niche modeling to the fossil record. *Paleobiology* **41**:226–244.
- Napier, J. D., G. de Lafontaine, K. D. Heath, and F. S. Hu. 2019. Rethinking long-term vegetation dynamics: multiple glacial refugia and local expansion of a species complex. *Ecography* 42:1056–1067.
- Nógues-Bravo, D. 2009. Predicting the past distribution of species climatic niches. *Global Ecology and Biogeography* 18:521–531.
- Nogués-Bravo, D., S. Veloz, B. G. Holt, J. Singarayer, P. Valdes, B. Davis, S. C. Brewer, J. W. Williams, and C. Rahbek. 2016. Amplified plant turnover in response to climate change forecast by Late Quaternary records. *Nature Climate Change* 6:1115–1119.
- Nyári, Á. S., and S. Reddy. 2013. Comparative phyloclimatic analysis and evolution of ecological niches in the scimitar babblers (Aves: Timaliidae: *Pomatorhinus*). *PLoS ONE* **8**:e55629.
- Owens, H. L., L. P. Campbell, L. L. Dornak, E. E. Saupe, N. Barve, J. Soberón, K. Ingenloff, A. Lira-Noriega, C. M. Hensz, and C. E. Myers. 2013.
 Constraints on interpretation of ecological niche models by limited environmental ranges on calibration areas. *Ecological Modelling* 263:10–18.
- Owens, H. L., V. Ribeiro, E. E. Saupe, M. E. Cobos, P. A. Hosner, J. C. Cooper, A. M. Samy, et al. 2020. Acknowledging uncertainty in evolutionary reconstructions of ecological niches. *Ecology and Evolution* 10:6967–6977.
- Parmesan, C., and G. Yohe. 2003. A globally coherent fingerprint of climate change impacts across natural systems. *Nature* 421:37–42.
- Patzkowsky, M. E. 2017. Origin and evolution of regional biotas: a deep-time perspective. Annual Review of Earth and Planetary Sciences 45:471–495.
- Patzkowsky, M. E., and S. M. Holland. 2016. Biotic invasion, niche stability, and the assembly of regional biotas in deep time: comparison between faunal provinces. *Paleobiology* 42:359–379.
- Payne, J. L., J. A. A. Aswad, C. Deutsch, P. M. Monarrez, J. L. Penn, and P. Singh. 2023. Selectivity of mass extinctions: patterns, processes, and future directions. *Cambridge Prisms: Extinction* 1:e12.
- Pearman, P. B., A. Guisan, O. Broennimann, and C. F. Randin. 2008. Niche dynamics in space and time. *Trends in Ecology and Evolution* 23: 149–158.
- Pecl, G. T., M. B. Araujo, J. D. Bell, J. Blanchard, T. C. Bonebrake, I.-C. Chen, T. D. Clark, et al. 2017. Biodiversity redistribution under climate change: impacts on ecosystems and human well-being. *Science* 355:eaai9214.
- Peters, S. E., J. M. Husson, and J. Czaplewski. 2018. Macrostrat: a platform for geological data integration and deep-time Earth crust research. *Geochemistry*, *Geophysics, Geosystems* 19:1393–1409.
- Peterson, A. T. 2011. Ecological niche conservatism: a time-structured review of evidence: ecological niche conservatism. *Journal of Biogeography* **38**:817–827.
- Peterson, A. T., J. Soberón, and V. Sánchez-Cordero. 1999. Conservatism of ecological niches in evolutionary time. *Science* 285:1265–1267.

- Phillips, S. J., M. Dudík, J. Elith, C. H. Graham, A. Lehmann, J. Leathwick, and S. Ferrier. 2009. Sample selection bias and presence-only distribution models: implications for background and pseudo-absence data. *Ecological Applications* 19:181–197.
- Pilowsky, J. A., R. K. Colwell, C. Rahbek, and D. A. Fordham. 2022a. Processexplicit models reveal the structure and dynamics of biodiversity patterns. *Science Advances* 8:eabj2271.
- Pilowsky, J. A., S. Haythorne, S. C. Brown, M. Krapp, E. Armstrong, B. W. Brook, C. Rahbek, and D. A. Fordham. 2022b. Range and extinction dynamics of the steppe bison in Siberia: a pattern-oriented modelling approach. *Global Ecology and Biogeography* 31:2483–2497.
- Pinilla-Buitrago, G. E., T. Escalante, A. Gutiérrez-Velázquez, P. Reyes-Castillo, and O. R. Rojas-Soto. 2018. Areas of endemism persist through time: a palaeoclimatic analysis in the Mexican Transition Zone. *Journal of Biogeography* 45:952–961.
- Prates, I., A. T. Xue, J. L. Brown, D. F. Alvarado-Serrano, M. T. Rodrigues, M. J. Hickerson, and A. C. Carnaval. 2016. Inferring responses to climate dynamics from historical demography in neotropical forest lizards. *Proceedings of the National Academy of Sciences USA* 113:7978–7985.
- Precht, W., and R. Aronson. 2004. Climate flickers and range shifts of reef corals. Frontiers in Ecology and the Environment 2:307–314.
- Purcell, C. K. Q., and A. L. Stigall. 2021. Ecological niche evolution, speciation, and feedback loops: investigating factors promoting niche evolution in Ordovician brachiopods of eastern Laurentia. *Palaeogeography, Palaeoclimatology, Palaeoecology* 578:110555.
- Pyron, R. A., and F. T. Burbrink. 2009. Can the tropical conservatism hypothesis explain temperate species richness patterns? An inverse latitudinal biodiversity gradient in the New World snake tribe Lampropeltini. *Global Ecology and Biogeography* 18:406–415.
- Qiao, H., E. E. Saupe, J. Soberón, A. T. Peterson, and C. E. Myers. 2016. Impacts of niche breadth and dispersal ability on macroevolutionary patterns. *American Naturalist* 188:149–162.
- Qiao, H., A. T. Peterson, L. Ji, and J. Hu. 2017. Using data from related species to overcome spatial sampling bias and associated limitations in ecological niche modelling. *Methods in Ecology and Evolution* 8:1804–1812.
- Qiao, H., X. Feng, L. E. Escobar, A. T. Peterson, J. Soberón, G. Zhu, and M. Papeş. 2019. An evaluation of transferability of ecological niche models. *Ecography* 42:521–534.
- Quental, T. B., and C. R. Marshall. 2010. Diversity dynamics: molecular phylogenies need the fossil record. *Trends in Ecology and Evolution* 25: 434–441.
- Quintero, I., M. A. Suchard, and W. Jetz. 2022. Macroevolutionary dynamics of climatic niche space. *Proceedings of the Royal Society B* 289:20220091.
- Quintero, I., M. J. Landis, W. Jetz, and H. Morlon. 2023. The build-up of the present-day tropical diversity of tetrapods. *Proceedings of the National Acad*emy of Sciences USA 120:e2220672120.
- **R Core Team**. 2014. *R: A language and environment for statistical computing*. R Foundation for Statistical Computing, Vienna, Austria.
- Reddin, C. J., P. S. Nätscher, Á. T. Kocsis, H.-O. Pörtner, and W. Kiessling. 2020. Marine clade sensitivities to climate change conform across timescales. *Nature Climate Change* 10:249–253.
- Reddin, C. J., M. Aberhan, N. B. Raja, and Á. T. Kocsis. 2022. Global warming generates predictable extinctions of warm- and cold-water marine benthic invertebrates via thermal habitat loss. *Global Change Biology* 28:5793– 5807.
- Rico, Y., M.Á. León-Tapia, M. Zurita-Solís, F. Rodríguez-Gómez, and S. G. Vásquez-Morales. 2021. Influence of Pleistocene climatic oscillations on the phylogeography and demographic history of endemic vulnerable trees (section *Magnolia*) of the Tropical Montane Cloud Forest in Mexico. *PeerJ* 9:e12181.
- Rindel, D. D., B. F. Moscardi, and S. I. Perez. 2021. The distribution of the guanaco (*Lama guanicoe*) in Patagonia during Late Pleistocene–Holocene and its importance for prehistoric human diet. *The Holocene* 31:644–657.
- Ritchie, J. C. 1995. Current trends in studies of long-term plant community dynamics. *New Phytologist* 130:469–494.
- Rivera, J. A., A. M. Lawing, and E. P. Martins. 2020. Reconstructing historical shifts in suitable habitat of *Sceloporus* lineages using phylogenetic niche modelling. *Journal of Biogeography* 47:2117–2128.

- Roberts, D. R., and A. Hamann. 2012. Predicting potential climate change impacts with bioclimate envelope models: a palaeoecological perspective. *Global Ecology and Biogeography* 21:121–133.
- Robin-Champigneul, F., J. Gravendyck, H. Huang, A. Woutersen, D. Pocknall, N. Meijer, G. Dupont-Nivet, R. H. J. Erkens, and C. Hoorn. 2023. Northward expansion of the southern-temperate podocarp forest during the early Eocene: palynological evidence from the NE Tibetan Plateau (China). *Review of Palaeobotany and Palynology* **316**:104914.
- Rocha, R. G., J. Gonçalves, P. Tarroso, P. Monterroso, and R. Godinho. 2022. Multiple lines of ecological evidence support ancient contact between the African wild dog and the dhole. *Frontiers in Ecology and Evolution* 10. https:// doi.org/10.3389/fevo.2022.803822.
- Rolland, J., and N. Salamin. 2016. Niche width impacts vertebrate diversification. Global Ecology and Biogeography 25:1252–1263.
- Rolland, J., D. Silvestro, D. Schluter, A. Guisan, O. Broennimann, and N. Salamin. 2018. The impact of endothermy on the climatic niche evolution and the distribution of vertebrate diversity. *Nature Ecology and Evolution* 2: 459–464.
- Romdal, T. S., M. B. Araújo, and C. Rahbek. 2013. Life on a tropical planet: niche conservatism and the global diversity gradient. *Global Ecology and Biogeography* 22:344–350.
- Rondinini, C., K. A. Wilson, L. Boitani, H. Grantham, and H. P. Possingham. 2006. Tradeoffs of different types of species occurrence data for use in systematic conservation planning. *Ecology Letters* 9:1136–1145.
- Rowan, J., J. M. Kamilar, L. Beaudrot, and K. E. Reed. 2016. Strong influence of palaeoclimate on the structure of modern African mammal communities. *Proceedings of the Royal Society B* 283:20161207.
- RStudio Team. 2020. RStudio: integrated development for R, version 2023.06.1 build 524, 'Mountain Hydrangea. RStudio, PBC, Boston, MA. URL http:// www.rstudio.com/, accessed on 7th July 2023.
- Rutrough, A., I. V. Widick, and W. T. Bean. 2019. Reconstruction of the historical range alters niche estimates in an endangered rodent. *Ecography* 42: 1742–1751.
- Sanz-Arnal, M., C. Benítez-Benítez, M. Miguez, P. Jiménez-Mejías, and S. Martín-Bravo. 2022. Are Cenozoic relict species also climatic relicts? Insights from the macroecological evolution of the giant sedges of *Carex* sect. Rhynchocystis (*Cyperaceae*). American Journal of Botany 109:115–129.
- Saupe, E. E., V. Barve, C. E. Myers, J. Soberón, N. Barve, C. M. Hensz, A. T. Peterson, H. L. Owens, and A. Lira-Noriega. 2012. Variation in niche and distribution model performance: the need for a priori assessment of key causal factors. *Ecological Modelling* 237–238:11–22.
- Saupe, E. E., J. R. Hendricks, A. Townsend Peterson, and B. S. Lieberman. 2014. Climate change and marine molluscs of the western North Atlantic: future prospects and perils. *Journal of Biogeography* 41:1352–1366.
- Saupe, E. E., H. Qiao, J. R. Hendricks, R. W. Portell, S. J. Hunter, J. Soberón, and B. S. Lieberman. 2015. Niche breadth and geographic range size as determinants of species survival on geological time scales. *Global Ecology and Biogeography* 24:1159–1169.
- Saupe, E. E., N. Barve, H. L. Owens, J. C. Cooper, P. A. Hosner, and A. T. Peterson. 2018. Reconstructing ecological niche evolution when niches are incompletely characterized. *Systematic Biology* 67:428–438.
- Saupe, E. E., A. Farnsworth, D. J. Lunt, N. Sagoo, K. V. Pham, and D. J. Field. 2019a. Climatic shifts drove major contractions in avian latitudinal distributions throughout the Cenozoic. *Proceedings of the National Academy of Sciences USA* 116:12895–12900.
- Saupe, E. E., C. E. Myers, A. Townsend Peterson, J. Soberón, J. Singarayer, P. Valdes, and H. Qiao. 2019b. Spatio-temporal climate change contributes to latitudinal diversity gradients. *Nature Ecology and Evolution* 3: 1419–1429.
- Sawyer, Y. E., S. O. MacDonald, E. P. Lessa, and J. A. Cook. 2019. Living on the edge: exploring the role of coastal refugia in the Alexander Archipelago of Alaska. *Ecology and Evolution* 9:1777–1797.
- Scheele, B. C., C. N. Foster, S. C. Banks, and D. B. Lindenmayer. 2017. Niche contractions in declining species: mechanisms and consequences. *Trends in Ecology and Evolution* 32:346–355.
- Scholl, J. P., and J. J. Wiens. 2016. Diversification rates and species richness across the Tree of Life. *Proceedings of the Royal Society B* 283:20161334.

- Slater, G. J., and L. J. Harmon. 2013. Unifying fossils and phylogenies for comparative analyses of diversification and trait evolution. *Methods in Ecol*ogy and Evolution 4:699–702.
- Smith, A. B., S. J. Murphy, D. Henderson, and K. D. Erickson. 2023. Including imprecisely georeferenced specimens improves accuracy of species distribution models and estimates of niche breadth. *Global Ecology and Biogeography* 32:342–355.
- **Soberón, J.** 2007. Grinnellian and Eltonian niches and geographic distributions of species. *Ecology Letters* **10**:1115–1123.
- Soberón, J., and M. Nakamura. 2009. Niches and distributional areas: concepts, methods, and assumptions. *Proceedings of the National Academy of Sciences* USA 106(Suppl. 2):19644–19650.
- Steffen, W., J. Rockström, K. Richardson, T. M. Lenton, C. Folke, D. Liverman, C. P. Summerhayes, et al. 2018. Trajectories of the Earth system in the Anthropocene. *Proceedings of the National Academy of Sciences USA* 115: 8252–8259.
- Stigall, A. L. 2012. Using ecological niche modelling to evaluate niche stability in deep time. *Journal of Biogeography* **39**:772–781.
- Stigall, A. L. 2014. When and how do species achieve niche stability over long time scales? *Ecography* 37:1123–1132.
- Stigall, A. L. 2015. Speciation: expanding the role of biogeography and niche breadth in macroevolutionary theory. Pp. 301–327 in E. Serrelli and N. Gontier, eds. *Macroevolution: explanation, interpretation and evidence*. Springer International Publishing, Cham, Switzerland.
- Stigall, A. L. 2023. A review of the Late Ordovician (Katian) Richmondian Invasion of eastern Laurentia. *Palaeogeography, Palaeoclimatology, Palaeoecology* 618:111520.
- Stigall Rode, A. L., and B. S. Lieberman. 2005. Using environmental niche modeling to study the late Devonian biodiversity crisis. Pp. 93–179 in D. J. Over, J. R. Morrow, and P. B. Wignall, eds. Understanding Late Devonian and Permian-Triassic biotic and climatic events: towards an integrated approach. Developments in Palaeontology and Stratigraphy, Vol. 20. Elsevier, Amsterdam.
- Sundaram, M., and A. B. Leslie. 2021. The influence of climate and palaeoclimate on distributions of global conifer clades depends on geographical range size. *Journal of Biogeography* 48:2286–2297.
- Svenning, J.-C., C. Fløjgaard, K. A. Marske, D. Nógues-Bravo, and S. Normand. 2011. Applications of species distribution modeling to paleobiology. *Quaternary Science Reviews* 30:2930–2947.
- Talukder, B., N. Ganguli, R. Matthew, G. W. vanLoon, K. W. Hipel, and J. Orbinski. 2022. Climate change-accelerated ocean biodiversity loss & associated planetary health impacts. *Journal of Climate Change and Health* 6: 100114.
- Tarkhnishvili, D., A. Gavashelishvili, and L. Mumladze. 2012. Palaeoclimatic models help to understand current distribution of Caucasian forest species. *Biological Journal of the Linnean Society* 105:231–248.
- Tilman, D., F. Isbell, and J. M. Cowles. 2014. Biodiversity and ecosystem functioning. Annual Review of Ecology, Evolution, and Systematics 45:471–493.
- Title, P. O., and D. L. Rabosky. 2019. Tip rates, phylogenies and diversification: what are we estimating, and how good are the estimates? *Methods in Ecology and Evolution* 10:821–834.
- Tóth, A. B., S. K. Lyons, W. A. Barr, A. K. Behrensmeyer, J. L. Blois, R. Bobe, M. Davis, et al. 2019. Reorganization of surviving mammal communities after the end-Pleistocene megafaunal extinction. *Science* 365:1305–1308.
- Tricco, A. C., E. Lillie, W. Zarin, K. K. O'Brien, H. Colquhoun, D. Levac, D. Moher, M. D. Peters, T. Horsley, and L. Weeks. 2018. PRISMA extension for scoping reviews (PRISMA-ScR): checklist and explanation. *Annals of Internal Medicine* 169:467–473.
- Valentine, J. W. 1969. Niche diversity and niche size patterns in marine fossils. *Journal of Paleontology* 43:905–915.
- van Bodegom, P. M., J. C. Douma, and L. M. Verheijen. 2014. A fully traitsbased approach to modeling global vegetation distribution. *Proceedings of the National Academy of Sciences USA* 111:13733–13738.
- van der Plas, F. 2019. Biodiversity and ecosystem functioning in naturally assembled communities. *Biological Reviews* 94:1220–1245.
- Varela, S., J. Lobo, and J. Rodríguez. 2010. Were the Late Pleistocene climatic changes responsible for the disappearance of the European spotted hyena

populations? Hindcasting a species geographic distribution across time. *Quaternary Science Reviews* **29**:2027–2035.

- Varela, S., J. M. Lobo, and J. Hortal. 2011. Using species distribution models in paleobiogeography: a matter of data, predictors and concepts. *Palaeogeography, Palaeoclimatology, Palaeoecology* **310**:451–463.
- Veloz, S. D., J. W. Williams, J. L. Blois, and F. He. 2012. No-analog climates and shifting realized niches during the late Quaternary: implications for 21stcentury predictions by species distribution models. *Global Change Biology* 18: 1698–1713.
- Vieites, D. R., S. Nieto-Román, and D. B. Wake. 2009. Reconstruction of the climate envelopes of salamanders and their evolution through time. *Proceedings of the National Academy of Sciences USA* 106:19715–19722.
- Villavicencio, N. A., D. Corcoran, and P. A. Marquet. 2019. Assessing the causes behind the Late Quaternary extinction of horses in South America using species distribution models. *Frontiers in Ecology and Evolution* 7. https://doi.org/10.3389/fevo.2019.00226.
- Wagner, P. J. 1995. Stratigraphic tests of cladistic hypotheses. Paleobiology 21: 153–178.
- Wang, Y., W. Porter, P. D. Mathewson, P. A. Miller, R. W. Graham, and J. W. Williams. 2018. Mechanistic modeling of environmental drivers of woolly mammoth carrying capacity declines on St. *Paul Island*. Ecology 99:2721–2730.
- Wang, Y., C. Widga, R. W. Graham, J. L. McGuire, W. Porter, D. Wårlind, and J. W. Williams. 2021. Caught in a bottleneck: habitat loss for woolly mammoths in central North America and the ice-free corridor during the last deglaciation. *Global Ecology and Biogeography* 30:527–542.
- Wang, Y., S. Pineda-Munoz, and J. L. McGuire. 2023. Plants maintain climate fidelity in the face of dynamic climate change. *Proceedings of the National Academy of Sciences USA* 120:e2201946119.
- Waterson, A. M., D. N. Schmidt, P. J. Valdes, P. A. Holroyd, D. B. Nicholson, A. Farnsworth, and P. M. Barrett. 2016. Modelling the climatic niche of turtles: a deep-time perspective. *Proceedings of the Royal Society B* 283: 20161408.
- Wehi, P. M., M. P. Cox, T. Roa, and H. Whaanga. 2018. Human perceptions of megafaunal extinction events revealed by linguistic analysis of indigenous oral traditions. *Human Ecology* 46:461–470.
- Wei, D., I. C. Prentice, and S. P. Harrison. 2020. The climatic space of European pollen taxa. *Ecology* 101:e03055.
- Wendt, J. A. F., D. B. McWethy, C. Widga, and B. N. Shuman. 2022. Largescale climatic drivers of bison distribution and abundance in North America since the Last Glacial Maximum. *Quaternary Science Reviews* 284: 107472.
- Werneck, F. P., C. Nogueira, G. R. Colli, J. W. Sites Jr., and G. C. Costa. 2012. Climatic stability in the Brazilian cerrado: implications for biogeographical connections of South American savannas, species richness and conservation in a biodiversity hotspot. *Journal of Biogeography* 39:1695–1706.
- Westgate, M. J. 2019. revtools: an R package to support article screening for evidence synthesis. *Research Synthesis Methods* 10:606–614.
- Whaanga, H., P. Wehi, M. Cox, T. Roa, and I. Kusabs. 2018. Māori oral traditions record and convey indigenous knowledge of marine and freshwater resources. New Zealand Journal of Marine and Freshwater Research 52:487–496.
- Wickham, H. 2023. stringr: simple, consistent wrappers for common string operations. R package version 1.5.0, https://CRAN.R-project.org/package= stringr, accessed 29 November 2024.
- Wickham, H., R. François, L. Henry, K. Müller, and D. Vaughan. 2023a. dplyr: a grammar of data manipulation. R package version 1.1.2, https://CRAN.Rproject.org/package=dplyr, accessed 29 November 2024.
- Wickham, H., D. Vaughan, and M. Girlich. 2023b. tidyr: tidy messy data. R package version 1.3.0, https://CRAN.R-project.org/package=tidyr, accessed 29 November 2024.
- Wiens, J. J. 2011. The niche, biogeography and species interactions. *Philosophical Transactions of the Royal Society B* 366:2336–2350.
- Wiens, J. J., and C. H. Graham. 2005. Niche conservatism: integrating evolution, ecology, and conservation biology. *Annual Review of Ecology, Evolution,* and Systematics 36:519–539.
- Wiens, J. J., D. D. Ackerly, A. P. Allen, B. L. Anacker, L. B. Buckley, H. V. Cornell, E. I. Damschen, et al. 2010. Niche conservatism as an emerging

principle in ecology and conservation biology. *Ecology Letters* 13: 1310–1324.

- Wieringa, J. G., M. R. Boot, M. V. Dantas-Queiroz, D. Duckett, E. M. Fonseca, H. Glon, N. Hamilton, et al. 2020. Does habitat stability structure intraspecific genetic diversity? It's complicated... Frontiers of Biogeography 12. https://doi.org/10.21425/F5FBG45377.
- Williams, J. W., B. N. Shuman, T. Webb III, P. Bartlein, and P. L. Leduc. 2004. Late-Quaternary vegetation dynamics in North America: scaling from taxa to biomes. *Ecological Monographs* 74:309–334.
- Williams, J. W., H. M. Kharouba, S. Veloz, M. Vellend, J. McLachlan, Z. Liu, B. Otto-Bliesner, and F. He. 2013. The ice age ecologist: testing methods for reserve prioritization during the last global warming. *Global Ecology and Biogeography* 22:289–301.
- Williams, J. W., E. C. Grimm, J. L. Blois, D. F. Charles, E. B. Davis, S. J. Goring, R. W. Graham, et al. 2018. The Neotoma Paleoecology Database, a

multiproxy, international, community-curated data resource. *Quaternary Research* 89:156–177.

- Williams, J. W., T. L. Spanbauer, P. D. Heintzman, J. Blois, E. Capo, S. J. Goring, M.-E. Monchamp, et al. 2023. Strengthening global-change science by integrating aeDNA with paleoecoinformatics. *Trends in Ecology and Evolution* 38:946–960.
- Wilson, O. J., F. E. Mayle, R. J. Walters, D. V. Lingner, and A. C. Vibrans. 2021. Floristic change in Brazil's southern Atlantic Forest biodiversity hotspot: from the Last Glacial Maximum to the late 21st century. *Quaternary Science Reviews* 264:107005.
- Yeakel, J. D., M. M. Pires, L. Rudolf, N. J. Dominy, P. L. Koch, P. R. Guimarães, and T. Gross. 2014. Collapse of an ecological network in ancient Egypt. *Proceedings of the National Academy of Sciences USA* 111:14472–14477.
- Zhang, Q., R. H. Ree, N. Salamin, Y. Xing, and D. Silvestro. 2022. Fossil-informed models reveal a boreotropical origin and divergent evolutionary trajectories in the walnut family (Juglandaceae). *Systematic Biology* 71:242–258.